

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Duffy, John Michael; Heinemann, Frank

# Working Paper Central bank reputation, cheap talk and transparency as substitutes for commitment: Experimental evidence

SFB 649 Discussion Paper, No. 2016-053

**Provided in Cooperation with:** Collaborative Research Center 649: Economic Risk, Humboldt University Berlin

*Suggested Citation:* Duffy, John Michael; Heinemann, Frank (2016) : Central bank reputation, cheap talk and transparency as substitutes for commitment: Experimental evidence, SFB 649 Discussion Paper, No. 2016-053, Humboldt University of Berlin, Collaborative Research Center 649 - Economic Risk, Berlin

This Version is available at: https://hdl.handle.net/10419/148889

### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU

# SFB 649 Discussion Paper 2016-053

# Central Bank Reputation, Cheap Talk and Transparency as Substitutes for Commitment: Experimental Evidence

John Duffy \* Frank Heinemann\*<sup>2</sup>



\* University of California, United States of America
 \*<sup>2</sup> Technische Universität Berlin, Germany

This research was supported by the Deutsche Forschungsgemeinschaft through the SFB 649 "Economic Risk".

http://sfb649.wiwi.hu-berlin.de ISSN 1860-5664

SFB 649, Humboldt-Universität zu Berlin Spandauer Straße 1, D-10178 Berlin



# **Central Bank Reputation, Cheap Talk and Transparency as Substitutes for Commitment: Experimental Evidence\***

John Duffy<sup>a</sup> and Frank Heinemann<sup>b</sup>

This Draft: 2 December 2016

**Abstract:** We implement a repeated version of the Barro-Gordon monetary policy game in the laboratory and ask whether reputation serves as a substitute for commitment, enabling the central bank to achieve the efficient Ramsey equilibrium and avoid the inefficient, time-inconsistent one-shot Nash equilibrium. We find that reputation is a poor substitute for commitment. We then explore whether central bank cheap talk, policy transparency, both cheap talk and policy transparency or economic transparency yield improvements in the direction of the Ramsey equilibrium under the discretionary policy regime. Our findings suggest that these mechanisms have only small or transitory effects on welfare. Surprisingly, the real effects of supply shocks are better mitigated by a commitment regime than by any discretionary policy. Thus, we find that there is no trade-off between flexibility and credibility.

Keywords: Monetary Policy, Repeated Games, Central Banking, Commitment, Discretion, Cheap Talk, Transparency, Experimental Economics.

JEL Codes: C92, D83, E52, E58.

<sup>\*</sup> We acknowledge financial support for this project from the Dietrich School of Arts and Sciences and from the Deutsche Forschungsgemeinschaft (DFG) through CRC 649 "Economic Risk" and Priority Program 1578 "Financial Market Imperfections and Macroeconomic Performance"

<sup>&</sup>lt;sup>a</sup> Department of Economics, University of California, Irvine, California, email: duffy@uci.edu

<sup>&</sup>lt;sup>b</sup> Department of Economics, Technische Universität Berlin, DE-10623 Berlin, Germany, email: frank.heinemann@tu-berlin.de

### 1. Introduction

Should central bankers commit to a consistent monetary policy or should they be afforded discretion to alter monetary policy depending on current circumstances? This question, as first posed by Kydland and Prescott (1977) and elaborated upon by Barro and Gordon (1983ab) continues to be debated in discussions of monetary policy design. On the one hand, the ability to use monetary policy to flexibly respond to various economic shocks as they arise is the main argument in favor of discretionary policy (e.g., Blinder (1999)). On the other hand, it is well known that the ability to commit to a consistent policy course of action or rule can yield welfare improvements over a pure discretionary policy regime through the effect that the commitment policy has on private sector expectations (e.g., Taylor (1999)). The latter argument hinges on the reputation that central bankers can achieve from consistently applying a low inflation monetary policy and thus envisions a repeated game setting between the policy maker and the private sector.

In this paper we implement a version of this repeated policy game due to Barro and Gordon in the laboratory with paid human subjects playing the role of the central banker and the private sector. We have several aims in mind. First, we wish to explore whether reputational considerations can serve as a substitute for commitment in a repeated game setting where central bankers lack a commitment device and are free to alter monetary policy each period conditioning on realizations of economic shocks. We compare that flexible, discretionary regime with a commitment regime where central bankers have the ability to precommit to a course for monetary policy in advance of the formation of private sector expectations, but no flexibility for stabilizing the economy.<sup>1</sup> Our experimental findings from these two regimes reveal that reputation is indeed a poor substitute for commitment in that inflation is higher and welfare is lower in the discretionary environment as compared with the commitment regime. Given that finding, our second goal is to evaluate various mechanisms that have been suggested to be useful for overcoming the temptation to succumb to high inflation time-consistent policy making in the repeated game discretionary environment.

<sup>&</sup>lt;sup>1</sup> A key advantage of conducting a laboratory experiment is that we are able to implement a commitment regime as a theoretical benchmark, while in more natural (i.e., non-laboratory) settings such commitment devices may have credibility problems.

In particular, we explore the role of cheap talk, policy transparency, both cheap talk and policy transparency, and finally economic transparency in overcoming the inflationary bias under discretionary policy. Many of these mechanisms have been studied in the context of Barro-Gordon type monetary policy game set-ups (see, Geraats (2002, 2014) for surveys) and thus the environment we study is appropriate for an analysis of the effectiveness of such mechanisms. We find that of the various mechanisms we study, cheap talk alone results in some welfare improvement relative to the baseline discretionary environment, but the benefits of such a mechanism appear to decline with experience. In the end, we conclude that none of the various mechanisms we examine comes close to achieving the welfare levels of the commitment regime. These findings suggest that there are real welfare-reducing consequences to discretionary monetary policy.

We adopt an experimental approach for several reasons. First and most importantly, the laboratory provides us with the control necessary to properly identify whether various monetary policy regimes, e.g., commitment versus various discretionary regimes matter, for monetary policy choices, private sector expectations and welfare. The control of the laboratory enables us to choose the timing of moves and the information available so as to closely match the theory being tested in a manner that is not possible using non-experimental field data. Second, the indefinitely repeated environments we study may admit multiple equilibria so that equilibrium selection becomes an issue. As Lucas (1986, p. 237) has argued, in settings with multiple equilibria, "it is hard to see what can advance the discussion short of assembling a collection of people, putting them in the situation of interest and seeing what they do." Finally, it is of interest to assess whether the theoretical predictions of the model are robust to agents who may depart in certain respects from the rational choice ideal. Our experiment makes use of student subjects to play the role of both central bankers and private sector agents. While ideally, we would have real central bankers make monetary policy choices in our experiment, there are good reasons to think that our experimental findings nevertheless remain externally valid and relevant to the discussion of actual central bank practice. In particular, there is evidence that student subjects' behavior is often not distinct from that of "professional" subjects (see, e.g. Frechette (2015)). Additionally, the literature on monetary policy emphasizes that central

bankers respond to monetary incentives (Walsh 1995), and our human subject central bankers are also incentivized to achieve good monetary policy outcomes in line with this view. Finally, while our precise quantitative experimental findings might be specific to the laboratory environment and to chosen parameter values (e.g., the discount rate), the *qualitative* results from comparing treatments with each other and with equilibria are more likely to be robust and externally valid.

The rest of the paper is organized as follows. Section 2 reviews the related literature. Section 3 outlines the one-shot and repeated Barro-Gordon model that we implement in the laboratory. Section 4 describes our experimental design and hypotheses and Section 5 reports on the main findings from our experiment. Finally, Section 6 concludes.

## 2. Related Literature

Three prior experiments have been conducted that are related to this paper. Van Huyck et al. (1995, 2001) implement a "peasant-dictator" game in the laboratory. In this two-player game, players are randomly paired, with one being designated as the peasant and the other the dictator. Pairs interact in an indefinitely repeated game that continues from one round to the next with probability 5/6. Peasants decide on how much of their endowment of beans to eat or plant (invest) yielding new beans in the next period (if there is a next period). Dictators tax production and can either pre-commit to a tax rate in advance of the peasant's investment (commitment regime) or decide on a tax ex-post, after investments have been made but prior to their realization (discretionary regime). The authors vary the endowments of the peasants and the interest rate earned on investments.<sup>2</sup> They report that reputation is an imperfect substitute for commitment and that efficiency under discretion is positively associated with the interest rate earned on investment.

Arifovic and Sargent (2003) implement an experimental version of the Kydland-Prescott model using a design that is similar to our own. In their study subjects are divided up between policymakers and private

 $<sup>^{2}</sup>$  They report that while they tried a treatment where the dictator made cheap talk announcements of intended tax rates (as we do here) they were dissatisfied with the results of this treatment and dropped it from their analysis.

sector forecasters with one policymaker and 3-5 private sector forecasters in each indefinitely lived match. Private sector forecasters move first: their objective is to correctly forecast next period's inflation. These inflation expectations then enter into a Phillips curve relation that determines the extent to which unemployment departs from its natural rate. The central bank moves second. It has noisy control over the actual inflation rate and seeks to minimize its expected loss from the equal weighted sum of the square of the unemployment and inflation rates. Arifovic and Sargent study only a discretionary regime and their treatment variables primarily consist of changes in the variance of shocks to the Phillips curve and inflation setting policy rule. By contrast with Van Huyck et al. they report that subjects do learn to coordinate on the first best Ramsey equilibrium consistent with a commitment regime, despite operating in a pure discretionary environment – that is they find that reputation does work as a substitute for commitment. In particular, they report that in three fourths of their sessions, policymakers eventually learn the Ramsey zero-inflation policy and stay with that policy for some time, though in several of the economies they report some "backsliding" toward the less efficient Nash equilibrium associated with one-shot pure discretionary regime after the Ramsey equilibrium had been achieved and sustained for some time.

Arifovic (2014) studies a version of the Kydland-Prescott model found in Arifovic et al. (2010) where the central bank makes cheap talk announcements about inflation in advance of private sector expectation formation and the private sector consists either of non-believers, or in a second treatment, non-believers and believers. The only experimental (human subjects) in this study are the non-believers. Both the central bank and the believer private sector agents are automated robot players who either learn over time in an evolutionary manner (the central bank) or blindly follow the central bank's pronouncements (the private sector believers). Arifovic's main finding is that with non-believers (human subjects) only or non-believers and believers (both human and robot private sector players) actual inflation levels lie below the one-shot Nash equilibrium prediction, though inflation is more volatile in the nonbeliever, humans-only, treatment.

Our experiment complements and builds upon these earlier experimental studies but differs from them in several respects. First, the Barro-Gordon (1983ab) model that we implement in the laboratory differs from the tax policy focus of the Van Huyck et al. study and differs in certain timing respects from the Kydland and Prescott model of monetary policy studied by Arifovic and Sargent (2003) and Arifovic (2014). For example, in the model we study, the central banker always learns inflation forecasts *in advance* of setting policy whereas in the Arifovic-Sargent study private forecasts are only known to policymakers ex-post. Also differently from Arifovic and Sargent, both policymakers and forecasters in our study are fully informed of the model economy, and subjects can play both roles over the course of an experimental session. By contrast with Arifovic (2014), we have no robot (or automated) players. Similar to Van Huyck et al. we study the case of both commitment and discretion. However we go beyond these two policy regimes and examine choices under the discretionary regime when 1) the policymaker can engage in preplay cheap talk about intended policy choices as in Arifovic (2014) but with real human subjects making those announcements; 2) policy choices are made transparent to the private sector at the end of each period, 3) there is both pre-game cheap talk and ex-post monetary policy transparency and 4) there is a regime of economic transparency where the private sector is informed about an economically relevant supply shock prior to forming their expectations (the policymaker is always informed of this shock in advance of setting policy). Thus our experiment goes beyond a comparison of repeated discretionary decision-making versus commitment and begins the important work of evaluating a number of non-reputation-based mechanisms by which it is thought that central bankers might overcome the inflation bias that is possible in the repeated but discretionary environments in which they operate.

Our paper is also related to experimental work involving the Stackelberg leadership model as our game always involves sequential moves though our treatments vary whether the first mover is the central bank or the private sector. Experimental research suggests that a sequential move, Stackelberg leadership structure yields higher welfare than does the comparable, simultaneous move (Cournot) model –see, e.g., Offerman et al. (2001). However, that study and other sequential move experimental games are not typically studied in indefinite-horizon environments as in this paper.

Cheap talk has also been explored by experimentalists as a mechanism for solving equilibrium coordination problems (see Crawford (1998) for a survey). Duffy and Feltovich (2006) report on experiments where the truthfulness of prior cheap talk messages (the extent of lying) can be evaluated by the receivers of those messages as in our policy transparency treatment. However, once again, that study and most other studies of pre-play communication have not been conducted in indefinitely repeated games where communication might help solve coordination problems arising from folk-theorem results. An exception is Camera et al. 2011 who study indefinitely repeated prisoner's dilemma games where free-form or structured pre-play communication is allowed at various intervals in the supergame. Similar to our findings, Camera et al. (2011) report that such communication does not do very much to help subjects achieve the most efficient equilibrium possible as subjects use communication for both benevolent and deceptive purposes.

#### 3. The Model

The model economy we implement in the laboratory is a version of that used by Barro and Gordon (1983ab). We begin with the static version before moving to the repeated (dynamic) version. Within the static environment we consider first the case of discretion and then the case of commitment.

### 3.1 Static model

The unemployment rate, y, is determined according to a Lucas-style aggregate supply function

$$y = y_n - c(\pi - \pi^e) + u_n$$

where  $y_n$  denotes the non-accelerating inflation rate of unemployment (NAIRU),  $\pi$  denotes the time t inflation rate,  $\pi^e$  denotes private sector expectations of the inflation rate at time t, c is a constant and u is a mean zero random supply shock. This supply function can be derived from the expectations-augmented Phillips curve view of the inflation-output tradeoff and implies that output and unemployment deviate from their natural rates only in response to unanticipated inflation. The central bank's monetary policy consists

of its choice of m, denoting the rate of growth of the money supply, which determines the actual inflation rate according to:

$$\pi = m + v$$
,

where v is a mean zero policy disturbance term (e.g., due to changes in the velocity of money or an unanticipated demand shock). The model is closed under the assumption that the private sector has rational expectations, so that  $\pi^e = E(\pi)$  and that the central bank seeks to minimize the time *t* loss function

$$EL = E(b(\pi - \pi^*)^2 + (y - y^*)^2)$$

where *EL* denotes the expected loss,  $\pi^*$  is the central bank's desired inflation rate and  $y^* < y_n$  denotes the central bank's desired unemployment rate which is assumed to be smaller than the NAIRU.

In the discretionary regime, the private sector moves first forming their expectations for inflation,  $\pi^e$ . The central bank is informed of these expectations and takes them as given when minimizing *L* subject to the expressions for *y* and  $\pi$ . The central bank's reaction function, given the private sector's expectations for inflation,  $\pi^e$ , is given by:

$$m = \frac{b\pi^* + c(y_n - y^* + u) + c^2 \pi^e}{b + c^2}.$$
(1)

The private sector is assumed to have rational expectations about inflation and we distinguish whether or not the private sector is informed about supply shocks when forming those expectations. If the private sector cannot observe supply shocks, we have that  $\pi^e = E(\pi) = E(m) = \frac{b\pi^* + c(y_n - y^*) + c^2\pi^e}{b + c^2}$ , or that  $\pi^e = \pi^* + \frac{c}{b}(y_n - y^*)$ , which implies that the money supply in Nash equilibrium is:

$$m^{NE} = \pi^* + \frac{c}{b}(y_n - y^*) + \frac{c}{b + c^2}u = \pi^{NE} + \frac{c}{b + c^2}u.$$

Thus the policy choice in the Nash equilibrium (NE) of the one-shot discretionary environment involves an inflation rate,  $\pi^{NE}$ , that is greater than the desired level,  $\pi^*$ , by the amount  $\frac{c}{b}(y_n - y^*)$ , which reflects the

well-known inflationary bias of discretionary policy. The third and final term,  $\frac{c}{b+c^2}u$ , reflects the central bank's incentive for stabilizing employment by adjusting the money supply to supply shocks, u.

If the central bank releases its information about supply shocks to the private sector before the formation of expectations, then  $\pi^e = E(\pi|u) = E(m|u)$ , which implies  $\pi^e = \pi^* + \frac{c}{b}(y_n - y^* + u)$  and

$$m^{ET} = \pi^* + \frac{c}{b}(y_n - y^* + u) = \pi^{NE} + \frac{c}{b}u.$$

We refer to this regime as one of "economic transparency" (ET) and in our experiment we will compare welfare under economic transparency with welfare under the pure (no economic transparency) discretionary and commitment regimes. As is well known from Geraats (2002, 2014), a discretionary regime with economic transparency combines a lack of credibility with a lack of flexibility since any policy responses of monetary policy to supply shocks are perfectly foreseen and, therefore cannot affect employment. The inflation bias in the discretionary regime with economic transparency is the same as in the discretionary regime with economic transparency is the same as in the discretionary regime without economic transparency, but the monetary policy response to supply shocks, is larger when there is economic transparency, causing a higher variation of inflation rates without stabilizing employment. The reason why the money supply responds so strongly to supply shocks is due to timing: the CB responds to private sector expectations that have already responded to the supply shock. Consequently, the total impact of shocks on inflation gets magnified under economic transparency. This regime combines time-inconsistent levels of inflation with time-inconsistent responses to shocks.

We next consider the commitment regime. In this environment the central bank moves first and precommits to set *m* in advance of private sector expectation formation but may be able to condition this decision on realizations of the shock,  $u_t$ . The central bank assumes that the private sector forms rational expectations,  $\pi^e = E(\pi|m) = m$ . Thus in this setting, the central bank's minimization problem is:

$$\min_{m} E[b(m+\nu-\pi^{*})^{2} + (y_{n} - c(m+\nu-\pi^{e}) + u - y^{*})^{2}|u], \text{ s.t. } \pi^{e} = m.$$

The solution, which we refer to as the one-shot commitment (C) equilibrium is given by  $m^{C} = \pi^{*}$ .

Under commitment, the central bank cannot create surprise inflation. Hence, it cannot stabilize the real economy and so its best policy is to target the optimal inflation rate irrespective of the supply shock. Comparing this case to the Nash equilibrium under economic transparency, both problems of time inconstancy are solved by the CB's ability to commit: inflation is both stable and at the optimal level. In comparison to the discretionary regime without economic transparency, however, we observe that while there is no inflationary bias, as  $\pi^* < \pi^{NE}$ , the inability to stabilize the real sector under commitment causes welfare losses that can be avoided under discretion.<sup>3</sup> This reflects the well-known trade-off between credibility and flexibility of monetary policy. This trade-off can be mitigated in the repeated (dynamic game) model where the private sector can condition its behavior on the central bank's past responses to supply shocks, so that reputational considerations come into play. We now turn our attention to that setting.

### 3.2 Dynamic model

Our experiment implements a dynamic repeated-game version of the model described in the previous section. In each period of this dynamic game, the central bank has the option to exploit low inflationary expectations by surprising the private sector with an unexpected high inflation which reduces unemployment and raises welfare in that period. However, unexpected high inflation can trigger a rise in future expectations about inflation which is to the central bank's disadvantage. Thus, the central bank has an incentive to keep inflation low, in order to maintain low inflationary expectations. Whether or not long-term reputational considerations for low inflation dominate short-term welfare gains from surprise inflation depends on parameters and on the effect of current inflation on future expectations. The highest incentive to keep inflation low arises if expectations are driven up forever. Here, we focus on conditions under which the efficient outcome, where  $\pi_t^e = \pi^*$ , can be sustained as an equilibrium of the infinitely repeated game.

In the repeated game, the central bank's objective is to

<sup>&</sup>lt;sup>3</sup> Whether the benefit from a flexible policy response to such shocks outweighs the costs arising from the inflationary bias of discretionary policy depends, of course, on the parameterization of the model.

$$min_{\{m_t\}} E_t \sum_{t=0}^{\infty} \delta^t \left[ b(\pi_t - \pi^*)^2 + (y_t - y^*)^2 | u_t \right]$$

subject to the given processes for  $\pi_t$  and  $y_t$ . Depending on the information available to the private sector, e.g., whether they learn ex-post about the policy rule m(u), and provided that the discount factor  $\delta$  is sufficiently large, the Folk theorem for infinitely repeated games applies so that the set of equilibria ranges from the one-period discretionary Nash equilibrium where  $m_t = \pi^{NE} + \frac{c}{b+c^2}u_t$  and private sector expectations satisfy  $\pi_t^e = \pi^{NE}$  to the efficient "Ramsey" solution where the central bank sets  $m_t = \pi^* + \frac{c}{b+c^2}u_t$ , avoiding the inflation bias, because  $\pi_t^e = \pi^*$ , but at the same time having the flexibility to stabilize employment. As this environment involves a multiplicity of equilibria with no clear means of choosing from among this set of equilibria, a laboratory experiment can be informative as to which equilibria agents are likely to coordinate on and under what conditions.

Before turning to the experimental design, we will briefly derive the conditions for the Ramsey solution being equilibrium of the repeated game. We assume throughout that the private sector will learn inflation  $\pi_t$  and unemployment  $y_t$  at the end of the period. Knowing the Phillips curve, it can thereby deduce the supply shock  $u_t$ . We need to distinguish, though, whether or not the private sector will get informed about the actual policy  $m_t$  and thereby the transmission shock  $v_t$ .

If policy is transparent (as assumed by Barro and Gordon (1983b)), expectations in period t+1 can be conditioned on the actual relationship between  $m_t$  and  $u_t$ . The strongest incentive for the central bank to pursue the Ramsey rule is supported by a grim trigger strategy played by private sector agents in which their expectations are initially  $\pi_0^e = \pi^*$  in the first period and remain there, as long as the central bank follows the Ramsey rule. If, however, the central bank deviates from this policy in any period  $\tau$ , the private sector's expectations immediately jump towards the one-period Nash equilibrium  $\pi_t^e = \pi^{NE}$  for all  $t > \tau$ , and the best response for the central bank is to follow the Nash-equilibrium policy  $m^{NE}$  in all future periods. Thus, the central bank faces the trade-off between exploiting low expectations and raising employment for one period on the one hand and implementing the maximum equilibrium inflation bias for all future periods as the result. If the central bank deviates in say period 0, it should best respond to  $\pi_0^e = \pi^*$  and  $u_0$ , which yields  $m_0 = \pi^* + \frac{c}{b+c^2}(y_n - y^* + u_0)$  and gives rise to a welfare loss of

$$\begin{split} & L_0^D(u_0) \\ &= E \left[ b \left( \frac{c}{b+c^2} (y_n - y^* + u_0) + v_0 \right)^2 + \left( y_n - c \left( \frac{c}{b+c^2} (y_n - y^* + u_0) + v_0 \right) + u_0 - y^* \right)^2 \Big| u_0 \right] \\ &= b \left( \frac{c}{b+c^2} (y_n - y^* + u_0) \right)^2 + b \sigma_v^2 + \left( \frac{b}{b+c^2} (y_n - y^* + u_0) \right)^2 + c^2 \sigma_v^2 \text{ [skip this line in publication]} \\ &= \frac{b}{b+c^2} (y_n - y^* + u_0)^2 + (b+c^2) \sigma_v^2 \,. \end{split}$$

The expected welfare loss associated with a deviation from Ramsey is then given by  $L_0^D(u_0) + \sum_{t=1}^{\infty} \delta^t E(L^{NE})$ , where  $E(L^{NE})$  is the prior expected welfare loss in the one-period Nash-equilibrium:

$$\begin{split} E(L^{NE}) &= E\left[b\left(\frac{c}{b}(y_n - y^*) + \frac{c}{b + c^2}u + v\right)^2 + \left(y_n - c\left(\frac{c}{b + c^2}u + v\right) + u - y^*\right)^2\right] \\ &= \frac{b + c^2}{b}(y_n - y^*)^2 + \frac{c^2}{b + c^2}\sigma_u^2 + (b + c^2)\sigma_v^2 \;. \end{split}$$

This must be compared with the expected welfare loss if the central bank follows the Ramsey rule. In the first period, this loss is given by

$$L^{R}(u_{0}) = E\left[b\left(\frac{c}{b+c^{2}}u_{0}+v_{0}\right)^{2}+\left(y_{n}-c\left(\frac{c}{b+c^{2}}u_{0}+v_{0}\right)+u_{0}-y^{*}\right)^{2}\Big|u_{0}\right]$$
  
$$= b\left(\frac{c}{b+c^{2}}u_{0}\right)^{2}+b\sigma_{v}^{2}+\left(y_{n}-y^{*}+\frac{b}{b+c^{2}}u_{0}\right)^{2}+c^{2}\sigma_{v}^{2} \quad [skip this line in publication]$$
  
$$= (y_{n}-y^{*})^{2}+\frac{b}{b+c^{2}}(u_{0}^{2}+2u_{0}(y_{n}-y^{*}))+(b+c^{2})\sigma_{v}^{2}.$$

The period-0 expectation of future losses under Ramsey is

$$E(L^{R}) = E\left[b\left(\frac{c}{b+c^{2}}u+v\right)^{2} + \left(y_{n}-c\left(\frac{c}{b+c^{2}}u+v\right)+u-y^{*}\right)^{2}\right]$$
$$= (y_{n}-y^{*})^{2} + \frac{c^{2}}{b+c^{2}}\sigma_{u}^{2} + (b+c^{2})\sigma_{v}^{2}.$$

Thus, the central bank has *no* incentive to deviate, if and only if  $L_0^D(u_0) + \sum_{t=1}^{\infty} \delta^t E(L^{NE}) \ge L_0^R(u_0) + \sum_{t=1}^{\infty} \delta^t E(L^R)$ , which is equivalent to

$$\frac{b}{b+c^{2}}(y_{n}-y^{*}+u_{0})^{2}-(y_{n}-y^{*})^{2}-\frac{b}{b+c^{2}}\left(u_{0}^{2}+2u_{0}(y_{n}-y^{*})\right) \geq \sum_{t=1}^{\infty}\delta^{t}\left[-\frac{c^{2}}{b}(y_{n}-y^{*})^{2}\right]$$
$$\Leftrightarrow \frac{-c^{2}}{b+c^{2}}(y_{n}-y^{*})^{2} \geq \frac{-\delta}{1-\delta}\cdot\frac{c^{2}}{b}(y_{n}-y^{*})^{2} \Leftrightarrow (1-\delta)b \leq \delta(b+c^{2}) \Leftrightarrow \delta \geq \frac{b}{2b+c^{2}}.$$
(1)

This condition is necessary and sufficient for the Ramsey solution to be an equilibrium under transparent monetary policy.

If policy is intransparent, the private sector cannot perfectly infer whether an increase in inflation is due to the central bank's deviating from the Ramsey rule or to an unfortunate realization of the transmission shock  $v_t$ . Here, the parameter restrictions that support the Ramsey equilibrium depend on the distribution of both shocks. In our experiment, we will use uniform distributions with bounded support. This allows us to derive a sufficient condition under which the Ramsey solution is an equilibrium.<sup>4</sup>

Suppose v has a uniform distribution in  $[-\mu, \mu]$  and consider the following strategy of forecasters: Expectations start with Ramsey and switch to Nash forever from period t + 1 onwards, if  $\pi_t > \pi^* + \mu$ . As long as the central bank plays Ramsey, the probability of expectations switching to Nash is zero. However, the central bank may raise the money supply just enough to exploit the large marginal gains for reducing

<sup>&</sup>lt;sup>4</sup> Henckel et al. (2011) discuss this problem for a normally distributed shock and a welfare function that is linear in output. They assume that expectations switch to the one-period Nash equilibrium for one period if a certain test statistic indicates that the central bank has been cheating with some given probability. However, the test statistic is chosen arbitrarily and it is assumed that the central bank does not strategically game the test statistic. Under these conditions, the Ramsey solution cannot be sustained as equilibrium.

unemployment from high levels at the risk of a moderate probability of being punished in the future. If  $\mu$  is large, the probability of detection is small, provided that *m* exceeds  $\pi^*$  just slightly. The CB can hide behind the shock, which may provide an incentive for deviations from Ramsey. For deriving a sufficient condition that prevents such incentives, first note that the marginal gain from increasing employment in the current period is a concave function of the money supply due to the quadratic loss function. The marginal expected future loss stemming from the probability of being detected, however, is linear due to the uniform distribution of transmission shocks.

If the money supply rises to  $m \in (\pi^*, \pi^* + 2\mu)$ , the probability of being detected is  $prob(\pi > \pi^* + \mu | m) = \frac{m - \pi^*}{2\mu}$ .<sup>5</sup> The associated expected welfare loss in the current period, say  $\tau = 0$ , is

$$\begin{split} E(L|u_0,m) &= E[b(m+v-\pi^*)^2 + (y_n - c(m+v-\pi^*) + u_0 - y^*)^2 | u_0] \\ &= b(m-\pi^*)^2 + b\sigma_v^2 + c^2(m-\pi^*)^2 - 2c(m-\pi^*)(y_n - y^* + u_0) + (y_n - y^* + u_0)^2 + c^2\sigma_v^2 \text{ [skip]} \\ &= (b+c^2)(m-\pi^*)^2 - 2c(m-\pi^*)(y_n - y^* + u_0) + (y_n - y^* + u_0)^2 + (b+c^2)\sigma_v^2. \end{split}$$

Thus, the marginal expected gain from increasing *m* is  $\frac{-\partial E(L|u_0,m)}{\partial m} = 2c(y_n - y^* + u_0) - 2(b + c^2)(m - \pi^*)$ . The marginal expected loss is  $\frac{1}{2\mu}\sum_{t=1}^{\infty}\delta^t (E(L^{NE} - L^R)) = \frac{\delta}{1-\delta} \cdot \frac{c^2}{2\mu b}(y_n - y^*)^2$ . A marginal deviation from Ramsey does *not* pay off, if and only if

$$2c(y_n - y^* - u_0) \le \frac{\delta}{1 - \delta} \cdot \frac{c^2}{2\mu b} (y_n - y^*)^2 \Leftrightarrow 4\mu \ b(y_n - y^* - u_0) \le \frac{\delta}{1 - \delta} \cdot c(y_n - y^*)^2.$$

Clearly, the incentive for inflating the economy rises with larger supply shocks. Since we assume a bounded support, a sufficient condition preventing deviations from Ramsey is

<sup>&</sup>lt;sup>5</sup> If money supply is larger, the deviation from Ramsey will be detected for sure. Condition (1) ensures that this is not in the interest of the central bank.

$$4\mu b(y_n - y^* + u_{max}) \le \frac{\delta}{1 - \delta} \cdot c(y_n - y^*)^2$$

$$\Leftrightarrow (1 - \delta) 4\mu b(y_n - y^* + u_{max}) \le \delta c(y_n - y^*)^2 \qquad [skip this line in publication]$$

$$\Leftrightarrow 4\mu b(y_n - y^* + u_{max}) \le \delta [c(y_n - y^*)^2 + 4\mu b(y_n - y^* - u_{min})] \qquad [skip this line in publication]$$

$$\Leftrightarrow \delta \ge \frac{4\mu b(y_n - y^* + u_{max})}{c(y_n - y^*)^2 + 4\mu b(y_n - y^* + u_{max})},$$
(2)

where  $u_{max}$  is the largest possible realization of the adverse supply shock. In the experiment, we will make sure that Conditions (1) and (2) hold.

### 4. Experimental Design and Hypotheses

Our experimental design consists of six different treatments that vary in the timing of moves and in the information available to participants. However, across all treatments, a number of factors were held constant and we begin with this basic structure.

### 4.1 Baseline Design

Each session of a given treatment involved 20 subjects with no prior experience with this experiment. The experiment was conducted over networked computers and was programmed using the z-Tree software (Fischbacher 2007). At the start of each session, subjects were randomly divided up into two matching groups of size 10. Subjects in different matching groups never interacted with each other and thus each matching group (2 per session) consistutes an independent observation. A session for each matching group consists of a number of repeated games known as "sequences" with each sequence consisting of an unknown number of rounds.

At the start of each new sequence, subjects in each matching group were randomly divided up into two groups of size 5 and the composition of the group remained constant for all rounds of the sequence. Prior to play of the first round of each sequence, one member of each group was randomly selected to play the role of the central banker, known as the "type A" player, while the other four members of each group were

assigned to play the role of the private sector, known as "type B" players.<sup>6</sup> Subjects remained in the *same* role in all rounds of a given supergame. At the start of each new sequence, groups were randomly formed anew and the type A player was again randomly chosen from among the membership of the new group, so that there is turnover of central bankers in our environment.

To avoid triggering any pre-conceived notions of the proper role or choices to be made by each player type, we used neutral language and a neutral framing of the model. For instance we did not want players to imagine they were central bankers choosing an inflation rate because they might have used their own experience from the real economy to coordinate on a particular rate, and we wanted the incentive structure of the model to determine their decisions instead.

We told subjects to imagine that the two variables,  $y_t$  and  $\pi_t$ , stand for two "containers" holding varying amounts of water<sup>7</sup>. Subjects were instructed that at the start of each round t =1,2,..., container 1 (unemployment) held  $w_t$  "gallons" ("liters") of water where  $w_t$  was publicly known to be an i.i.d. random draw each period from a uniform distribution over the interval [120, 160]. The expected value,  $E[w_t] =$ 140, corresponds to the NAIRU,  $y_n$ , in the model while the supply shock  $u_t = w_t - E[w_t]$ . Thus, in our parameterization one can think of the adverse supply shock,  $u_t$ , as an i.i.d. random draw from a uniform distribution with support [-20, 20] and thus having mean 0 and standard deviation  $20/\sqrt{3} = 11.55$ . The initial amount of water in Container 1 (unemployment) thus consists of both the NAIRU and the adverse supply shock, i.e.,  $w_t = y_n + u_t$ . The timing of when or whether players learned the value of  $w_t$  is an important element of our treatments. Subjects were further instructed that Container 2 (inflation) was initially empty.

<sup>&</sup>lt;sup>6</sup> Our choice of having a single central banker and a larger "private sector" of four players follows the setup of Arifovic and Sargent (2003) and reflects that fact that the private sector is considerably larger than the government sector.

<sup>&</sup>lt;sup>7</sup> The idea of framing a monetary policy game in terms of targeting amounts of water or chips in a container has been used first by Engle-Warnick and Turdaliev (2010).

In our baseline, discretionary policy treatment, the timing of moves was as follows. The four type B players in each economy moved first each submitting a forecast,  $\pi_{i,t}^e$ , as to how many gallons (liters) of water would be in Container 2 at the end of round t. They did so without knowing the realized value of  $w_t$ , though they did know that  $w_t$  was an i.i.d. random draw from a uniform distribution on the interval [120, 160] and they were told that  $E[w_t] = 140$ . They were also informed about the player A's objective function (as described below), so they knew the Player A's (central bank's) target values for inflation,  $\pi^*$ , and unemployment,  $y^*$ . After all four Player Bs had made their forecasts, the computer program calculated the mean forecast  $\pi_t^e = \frac{1}{4} \sum_{i=1}^4 \pi_{i,t}^e$  for the economy/group and revealed this value to the group's Player A – this forecast corresponds to  $\pi_t^e$  in the model. Subjects were instructed that this average forecast value would be added to the amount of water that was already in Container 1, so that the amount of water in Container 1 now increased to  $w_t + \pi_t^e$ . Then, the player A alone in each group learned the value of both  $w_t$  and  $\pi_t^e$ and the sum  $w_t + \pi_t^e$ , representing the new total amount of water in Container 1. Player A was then instructed to "move" some amount  $m \in [0,80]$  of water from Container 1 to Container 2. This choice corresponds to the policy choice of  $m_t$  for period t. In the baseline discretionary treatment, Player Bs do not observe Player A' choice for  $m_t$  but it is public knowledge that  $m \in [0,80]$ . In addition, it was public knowledge to both player types that there was a random, uncontrolled flow of water,  $v_t$ , from Container 1 to Container 2, corresponding to the policy transmission shock of the theory. The value of  $v_t$  was publicly known to be an independent random draw each period from a uniform distribution having support [0, 40] and all subjects were instructed that  $E[v_t] = 20$ . Note that differently from the theory, the transmission shock is not mean zero; this choice was made because the policy action space was  $m \in [0,80]$ , and we did not want to have inflation  $\pi_t = m + v_t$  be negative. Player As do not observe the value of  $v_t$  until after they have chosen  $m_t$ .

At the end of each period t, the final amount of water in Container 1 is thus given by  $w_t + \pi_t^e - m_t - v_t$ , which correlates to the Phillips curve relationship, in which surprise inflation reduces unemployment. Setting the parameter of the Phillips curve c = -1, as we do in all treatments of our experiment, we have that the final amount of water in Container 1 corresponds to  $y_t = y_n + \pi_t^e - \pi_t + u_t$  while the final amount of water in Container 2 corresponds to  $\pi_t = m_t + v_t$ .

The final amounts of water in Containers 1 and 2 were revealed to all subjects in each economy of size 5 at the end of each period as these amounts determined the subjects' payoffs for the round. Specifically, each player type was incentivized to make choices consistent with the objective functions posited by the theory. Type A players were instructed that their point earnings were given by the formula:

Player A Points = 6000 - 2 (Final Container 1 amount - 120)<sup>2</sup> – (Final Container 2 amount - 40)<sup>2</sup>.

Thus, Player A's (central bankers) had as their policy objectives:  $y^* = 120$  and  $\pi^* = 40$  and the parameter *b* was set equal to ½. As noted above, the payoff function and the parameter choices for Player As were public knowledge to all participants as revealed in the written instructions. Type B players were instructed that their point earnings were given by the formula:

Player B Points =  $4000 - (\pi_{i,t}^e - \text{Final Container 2 amount})^2$ .

Thus, Player B's had the simpler task of just forecasting the value for  $\pi_t$ , the amount of water in Container 2 correctly.<sup>8</sup> For this reason, the maximum number of points that Player Bs could earn each round was 4,000, while Player A's, who had the more difficult decision, could earn a maximum of 6,000 points per round. These equations determining players' points were presented to both player types and for ease of understanding, subjects were given payoff tables showing how their choices would convert into points each

<sup>&</sup>lt;sup>8</sup> The realized levels of inflation and unemployment do not enter into the payoff functions of forecasters, only the payoff functions of the central bankers. We are assuming that inflation forecast accuracy is all that matters to forecasters; given accurate inflation forecasts these agent would be able to infer the level of unemployment (output) from knowledge of the model and solve any profit or utility maximizing problems that they faced. Restricting private sector agents to forming inflation forecasts alone (a "learning to forecast" design) also limits such agents' incentives to affect aggregate variables such as unemployment and inflation which they might otherwise have incentives to influence given the small group sizes of the economies that we study. Thus our restriction that private sector agents provide forecasts only is also consistent with standard assumptions of perfectly competitive behaviour by private sector agents.

round. The experimental instructions including these payoff tables and other details about what subjects were told are provided in the Appendix. The parameterization of the model is justified in section 4.3.

At the end of each round, all subjects are informed of the final amounts of water in the two containers corresponding to the realizations of  $y_t$  and  $\pi_t$  and their point earnings for the round as determined by the expressions given above. In addition, type B players learn the realization of  $w_t = y_n + u_t$ . As they were told at the outset that the mean value for  $w_t = 140 = y_n$ , in effect they learn the realization of the supply shock  $u_t$  at the end of the period. Whether type B players learn the central banker's choice for  $m_t$  depends on the treatment as detailed below

Subjects were instructed that at the end of each round that a six-sided die would be rolled to determine whether the sequence continued with another round; they were further instructed that if a six was rolled, the sequence ended but otherwise the sequence would continue with another round. This constant random continuation probability implements both discounting with factor  $\delta = \frac{5}{6}$  and the stationarity associated with an infinite horizon. Subjects were instructed that, depending on the time remaining in the session, a new indefinite sequence might begin.

Subjects were informed that at the end of the session (which could last up to three hours), two sequences would be randomly chosen from among all those played and subjects point earnings from the chosen sequences would be converted into cash at a known and fixed rate. In addition, at the start of each sequence subjects were endowed with 5,000 points to avoid negative payoffs; since two sequences were randomly drawn at the end, the endowment served as subjects' show-up payment.

### 4.2 Treatments

Our experiment consists of six different treatments that are intended to explore the role (if any) played by reputation, cheap talk and transparency (both policy and economic) on welfare in the repeated discretionary environment relative to an environment where central banks can pre-commit to monetary policy. The six treatments are:

1. **Discretionary policy**: The timing and moves for this baseline treatment are as described in Section 4.1. The private sector, type Bs, move first and are not informed of the realization of the supply shock  $u_t$  when forming their expectations,  $\pi_{i,t}^e$ . The policy maker observes  $u_t$  and  $\pi_t^e$  and then chooses  $m_t$ . Type Bs never learn the value of  $m_t$  or  $v_t$ , but they do learn the final amounts of water in each container,  $y_t$  and  $\pi_t$  and are informed of the values for  $u_t$  and  $\pi_t^e$  at the end of the period.

2. Commitment: In this treatment, the central banker, Type A moves first, observing the realization of the supply shock,  $u_t$  and then choosing  $m_t$  prior to the formation of inflationary expectations by the private sector. Thus the commitment environment we study provides the central banker with the ability to respond to supply shocks but also to pre-commit to an inflation policy for the period.

3. Cheap talk (CT): In this treatment, Type A players again move first, observing the value of the supply shock  $u_t$  and then choose a message of the form: "The amount of water I intend to move from Container 1 to Container 2 is \_\_." In the blank space they enter the value of  $m \in [0,80]$  that they want to signal to forecasters in that round. This message is sent to all Player Bs in their group. Then the Player Bs form their forecasts of inflation for the period,  $\pi_{i,t}^e$ . Recall that Player Bs understand that the final amount of water in Container 1,  $\pi_t = m_t + v_t$ , so the announcement concerning the intended choice for  $m_t$  can play a role in coordinating private sector inflationary expectations. To ensure that the message is understood to be cheap talk, subjects are further instructed that "it is up to Player A whether he or she moves as much water as previously announced. Player A can move the announced amount or more or less water."

4. Policy transparency (PT): This treatment has the same timing as the baseline discretionary treatment. The only difference is that the private sector Type B players learn the realizations of both  $m_t$  and  $v_t$  at the end of each round, thus making it transparent as to whether inflation was high (low) due to the transmission shock or due to the policymaker's choice. A transparent policy environment such as in this and the next treatment (cheap talk plus policy transparency) makes it easier to sustain the Ramsey solution as equilibrium of the repeated game.

5. Cheap talk and policy transparency (CT+PT): This treatment combines the cheap talk phase of the cheap talk treatment with the information revealed about monetary policy ( $m_t$  and  $v_t$ ) at the end of each round as in the policy transparency treatment. This treatment thus allows the private sector to evaluate the truthfulness of the central bank's cheap talk announcements providing a potentially more credible means by which the central bank can attempt to manage private sector expectations as compared with cheap talk or policy transparency by themselves.

6. Economic transparency (ET): In this treatment, the private sector (Player Bs) learn the value of the supply shock  $u_t$  at the same time that the central banker (Player A) learns it, and prior to forming expectations of inflation for the period. The timing and information is otherwise identical to that of the discretionary treatment.

## 4.3 Calibration and Interpretation

Parameter choices had to satisfy different requirements: subjects must be able to comprehend the relation between different variables, the different equilibria should be sufficiently differentiated, allowing us to detect treatment effects and potentially reject hypotheses, the parameters should allow for an interpretation as a monetary policy game, the continuation probability should allow for several sequences per session and also satisfy the conditions for the Ramsey solution to be an equilibrium. While the levels of the unemployment and inflation target  $y^*$  and  $\pi^*$  are mere normalizations, the other parameters affect the interpretation.

We suggest that a period in the experiment corresponds to 2 years, our guess regarding the length of time it takes for monetary policy to have an impact on inflation and unemployment. Thus our continuation probability choice of  $\delta = 5/6$  means that the expected duration of a supergame (or a CB policy regime) is 12 years and implies an annual discount rate of about 8 percent. We assume that the difference between

natural and efficient unemployment is 2 percent. As we set  $y_n = 140$  and  $y^* = 120$ , the difference between efficient and natural unemployment is 20, so we may think of units of y as equaling 0.1 percent unemployment. We suppose that a 1 percent increase in inflation leads to a .5 percent reduction in unemployment. Setting c = 1, thus, implies that units of  $\pi$  must correspond to .2 percent inflation. Since the standard deviation of supply shocks u is 11.55 and a unit of y is .1 percent unemployment, the standard deviation of supply shocks in the experiment is about 1.15 percent unemployment. The transmission shock, v, also has a standard deviation of 11.55 and since a unit of  $\pi$  is .2 percent inflation; the standard deviation of policy transmission shocks in the experiment is 2.3 percent inflation. The parameter choices with b = .5also imply that the inflation bias in the one-period Nash equilibrium is  $\frac{c}{b}(y_n - y^*) = 40$  and, since units of  $\pi$  correspond to .2 percent inflation, corresponds to 8 percent inflation. Finally, we note that the optimal coefficient attached to supply shocks in the one-period Nash equilibrium and in the Ramsey solution is  $\frac{c}{b+c^2} = \frac{2}{3}$ . Suppose that the economy is hit by a shock that would increase unemployment by 1% without further policy measures (u = 10). In our economy, the optimal response would be an increase of  $\pi$  by 6.67 corresponding to 1.33% inflation. This policy reduces the increase in unemployment to .67%.

Note that the condition for existence of a Ramsey equilibrium in the repeated game under policy transparency requires  $\delta \ge \frac{b}{2b+c^2} = 0.25$ . With  $\mu = 20$  and  $u_{max} = 20$ , the sufficient condition for the Ramsey solution to be an equilibrium in the regimes without policy transparency is  $\delta \ge \frac{4\mu b(y_n-y^*+u_{max})}{c(y_n-y^*)^2+4\mu b(y_n-y^*+u_{max})} = \frac{40*40}{(20)^2+40*40} = 0.8$ . Naturally, it is more challenging than the condition under policy transparency. Our discount factor of  $\delta = 5/6$  satisfies both requirements.

### 4.4 Experimental Hypotheses

Given our parameterization of the model, theory predictions are summarized in Table 1, which reports point predictions if all equilibria give the same predictions or upper and lower bounds for the range of equilibria. The commitment solution is for the CB to set  $m_t = m = 20$  for all t since  $E[v_t] =$  20, and therefore  $E[\pi_t] = m + E[v_t] = 40 = \pi^*$ . As laid out in Section 3.1, the CB should not respond to supply shocks, because any such response would be reflected in forecasts and, thus, not affect employment. It follows that inflation varies only with the transmission shock,  $Std[\pi] = Std[v] = 11.55$ , while the variance of unemployment is the sum of the variances of the two shocks, or  $Std[y]=\sqrt{Var[v] + Var[u]} = 16.33$ . Expectations should be rational and equal to  $\pi^*$ , so that the standard deviation of  $\pi_t^e - \pi_t$  should simply reflect the standard deviation of the policy transmission shock, Std[v] = 11.55. Expected CB welfare is given by

$$6000 - 2E((140 - 120 - u - v)^2) - E(v^2) = 5200 - 2Var(u) - 3Var(v) = 4533.3.$$

In the case of discretion, we have multiple equilibria ranging from the one-period Nash with an inflation bias of 40 to the Ramsey solution without inflation bias. In all equilibria, the CB responds to supply shocks and stabilizes employment with a coefficient of  $\partial m/\partial u = 2/3$ . The same coefficient applies to the central bank's response to inflation expectations, as given by the response function (1). Note however that in a Ramsey equilibrium forecasts are constant and, thus, there should be no detectable variation to which the central bank responds. As subjects are most likely not exactly in equilibrium, the best response of the central bank with respect to "near Ramsey" expectations might still be zero. In the Result section, we will analyze whether subjects respond in an optimal way to the actual decision of other players. Assuming rational expectations, the actual choices of *m* should range from  $\pi^* - E(v) + \frac{2}{3}u_{min} = 6.67$  under Ramsey to  $\pi^* + 40 + E(v) + \frac{2}{3}u_{max} = 73.33$ . Our subject central bankers can choose values of  $m \in [0,80]$ , which contains all of these vales in the interior, allowing us to test the point predictions associated with the most extreme equilibria.

Treat. Predict	Commitment	Discretion	Cheap Talk	Policy Transparency	CT + PT	Economic Transparency
Inflation bias	0	[0,40]	[0,40]	[0,40]	[0,40]	[0,40]
Response of $m_t$ to supply shock, $u_t$ .	0	2/3	2/3	2/3	2/3	2/3 2 *
Response of $m_t$ to exp inflation, $\pi_t^e$ .	n.a.	2/3	2/3	2/3	2/3	2/3
Response of $\pi_t^e$ to supply shock, $u_t$	0	n.a.	0	n.a.	0	2
St.Dev. of $\pi_t$	11.55	13.88	13.88	13.88	13.88	25.82
St. Dev. of $y_t$	16.33	12.17	12.17	12.17	12.17	16.33
St. Dev. of $(\pi_t^e - \pi_t)$	11.55	13.88	13.88	13.88	13.88	11.55
CB Welfare	4533	[3111, 4711]	[3111, 4711]	[3111, 4711]	[3111, 4711]	[2400, 4000]
Forecasters' payoffs	3867	3807	3807	3807	3807	3867

**Table 1: Equilibrium Predictions** 

Note: \* The response of m to the supply shock in ET should be 2/3 if the response to private expectations is controlled for. Since expectations respond to the supply shock as well, the total response of m to u should be 2.

The stabilization of employment raises the standard deviation of inflation to  $\operatorname{Std}[\pi] = \sqrt{\operatorname{Var}[v] + \frac{4}{9}\operatorname{Var}[u]} = 13.88$ , while it reduces the standard deviation of unemployment to  $\operatorname{Std}[y] = \sqrt{\operatorname{Var}[v] + \frac{1}{9}\operatorname{Var}[u]} = 12.17$ . Since inflation expectations are constant in any equilibrium, the standard deviation of  $\pi_t^e - \pi_t$  should equal the standard deviation of inflation.

In the Ramsey equilibrium, expected CB welfare is:

$$6000 - 2E\left(\left(140 - 120 - \frac{1}{3}u - v\right)^2\right) - E\left(\left(v + \frac{2}{3}u\right)^2\right) = 5200 - \frac{6}{9}\operatorname{Var}(u) - 3\operatorname{Var}(v) = 4711.1.$$

In the Nash equilibrium, however, expected welfare is lower, because of the inflation bias and is given by

$$6000 - 2E\left(\left(140 - 120 - \frac{1}{3}u - v\right)^2\right) - E\left(\left(40 + v + \frac{2}{3}u\right)^2\right) = 3600 - \frac{6}{9}\operatorname{Var}(u) - 3\operatorname{Var}(v) = 3111.1.$$

Notice that the discretionary regime admits a range of possible welfare values for the CB that includes the commitment welfare value in its interior.

Policy transparency, cheap talk, and policy transparency plus cheap talk have no impact on the set of equilibria. While cheap talk announcements should be ignored in equilibrium, they might transmit information about the supply shock to the private sector, which justifies testing the equilibrium prediction of expectations being unresponsive to supply shocks and announcements.

Under economic transparency, on the other hand, the private sector learns the supply shock prior to forming expectations. As laid out in Section 3.1, this raises the equilibrium coefficient by which money supply, inflation, and inflation expectations respond to supply shocks to c/b = 2 without stabilizing employment, which results in  $Std[\pi] = \sqrt{Var[v] + 4Var[u]} = 25.82$ , while Std[y] = 16.33 as in the commitment regime. Because the private sector can predict monetary policy responses to supply shocks, the standard deviation of  $\pi_t^e - \pi_t$  should simply reflect the policy transmission shock, Std[v] = 11.55. As in the baseline discretionary treatment, there is a set of equilibria, ranging from the one period Nash to a constrained efficient solution, in which there is no inflation bias, but no stabilization of employment either. Welfare in the one-period Nash is given by

 $6000-2E((140-120-u-v)^2)-E((40+v+2u)^2) = 3600-6Var(u) - 3Var(v) = 2400$ . In the most efficient equilibrium, welfare is

$$6000 - 2E((140 - 120 - u - v)^2) - E((v + 2u)^2) = 5200 - 6Var(u) - 3Var(v) = 4000.$$

Beside the point predictions from equilibria, we have some more fundamental hypotheses that can be divided up between those pertaining to CB behavior, those pertaining to the behavior of the private sector and aggregate outcomes involving both sets of actors.

### Central Bank Hypotheses

**Hypothesis 1**. *Repetition in the discretionary regime serves as a substitute for commitment regarding monetary policy.* 

By this we simply mean that in the repeated game the CB behaves as if s/he were operating under a commitment regime and (a) produces a similar average money supply. The counter-hypothesis is that a higher money supply arises from the inflation bias associated with the less efficient equilibria under discretion. (b) The second part of this hypothesis states that monetary responses to supply shocks are the same. Here, the counter-hypothesis comes from the positive coefficients in the equilibria of discretionary regimes.

**Hypothesis 2**. *The discretionary regimes with and without cheap talk, policy transparency, or economic transparency produce the same central bank behavior.* 

More precisely, we test, (a) whether these treatments lead to the same average money supply and (b) to the same central bank responses to expected inflation and supply shocks.

### Private sector hypotheses

**Hypothesis 3**: Average private sector forecasts of inflation are unbiased and the distribution of forecast errors reflects the unpredictable fluctuations of inflation only.

This hypothesis follows from the assumption of rational expectations. We will test whether biases are null and, if not, whether they differ across treatments.

Hypothesis 4: The distribution of individual forecasts around the average is the same across treatments.

Theory predicts that B-players make the most precise forecasts and earn the highest payoffs in the commitment treatment and under economic transparency, and lower (but equal) payoffs in the other discretionary treatments. Besides a treatment-specific bias, individual forecast errors reflects how much information about economic shocks and monetary policies the B-players have when they make their

forecasts. We will test whether the standard deviation of average forecast errors is the same across treatments and how dispersed individual expecations are. While in equilibrium all agents hold the same expectations, in the experiment expectations will differ across subjects. These differences may be related to the treatment, even if the average forecast errors are the same across treatments. Related to this, the payoffs for type-B players might differ across treatments for three reasons: (i) there may be a systematic bias (cf. Hypothesis 3), (ii) average forecast errors may be more volatile in some treatments than in others (Std. of  $\pi_t^e - \pi_t$ ) and (iii) individual forecasts may be more or less dispersed (Std. of  $\pi_t^i - \pi_t^e$ ).

# **Hypothesis 5**: *Except for the economic transparency regime, inflation forecasts do not respond to supply shocks.*

In the baseline discretionary treatment and for policy tgransparency, B-players do not receive any information that might eventually depend on supply shocks. In treatments with cheap-talk, the announcements should be ignored and under commitment, monetary policy should not respond to supply shocks. The counter-hypothesis (that forecasts are affected by supply shocks under commitment or cheap talk) arises from possible information transmission in these treatments, if either cheap-talk announcements or money supply under commitment respond to these shocks.

Finally, we have some hypotheses regarding aggregate outcomes:

**Hypothesis 6**: Under commitment and economic transparency, output volatility is larger than in the other discretionary regimes.

Here, we test whether output volatility is the same between ET and commitment, whether it is the same across the other treatments, and whether there are differnces between these two groups of treatments.

**Hypothesis 7**: Compared with discretionary treatments, the standard deviaton of inflation should be higher under economic transparency and lower under commitment.

**Hypothesis 8**: Repetition in the discretionary regime serves as a substitute for commitment regarding (central bank) welfare.

While the inflation bias in discretionary regimes may be higher than under commitment, the flexibility with which central banks can respond to supply shocks under discretion, may reduce employment fluctuations. Thus, there are two opposing effects for the final level of central bank welfare. This can be seen in Table 1, where equilibrium welfare under commitment is strictly in between the lowest and highest welfare levels associated with equilibria in discretionary treatments.

**Hypothesis 9**: Under economic transparency, central bank welfare is lower than under commitment or in the baseline discretionary treatment.

For economic transparency the welfare levels in all equilibria are lower than under commitment. The only welfare relevant difference between economic transparency and the baseline discretionary treatment is that economic transparency does not allow for stabililizing output (in theory). This justifies our last hypothesis.

We generally use non-parametric tests based on average observations from a matching group to compare levels. We apply this conservative treatment of data, because behavior of different subjects from the same matching group need not be independent. For point predictions arising from theory, we use the two-sided Wilcoxon matched pairs test, for comparing different treatments, we use the two-sided Mann-Whitney Utest. More precise coefficient predictions are tested on the basis of confidence intervals from fixed effects regressions, where the unit of observation is matching group, sequence (supergame) number and subject ID.

## 4.5 Subjects, Sessions and Earnings

As noted in section 4.1 each experimental session consisted of 20 inexperienced subjects who were further divided up into two "matching groups" of size 10; the subjects in each matching group never interacted with one another and thus each matching group (2 per session) consistutes an independent observation.

Sessions were conducted at the Technical University of Berlin and the University of Pittsburgh. Specifically, two sessions of all 6 treatments were conducted in Berlin and two sessions of all 6 treatments were conducted in Pittsburgh. Therefore, we have 8 matching groups of 10 subjects (our observational unit) for each of our 6 treatments (4 groups from Berlin and 4 groups from Pittsburgh), for a total of  $8 \times 10 \times 6 = 480$  subjects. We did not find subtantial differences in behavior between the two subject populations (Berlin and Pittsburgh) and so in the analysis that follows we have pooled the data from all matching groups of a given treatment.

At the start of each treatment, subjects were read instructions aloud (see the Appendix for the sample instructions from the baseline discretionary treatment) and had to answer quiz questions before playing a number of supergames that constituted our experiment. Each session lasted 2-3 hours (subjects were always invited for 3 hours) and involved one or more open-ended sequences (supergames). Subjects were paid their earnings from 1 or a maximum of 2 supergames (the latter case applied if 2 or more supergames were played). They did not know in advance, which supergames were chosen for payoffs. Total payoffs ranged from USD \$13.30 to \$56.46.

### 5. Experimental Results

We report our experimental results as a number of different findings which address Hypotheses 1-9 as set forth in Section 4.4.

### 5.1 Money supply

Evidence in support of Finding 1 is presented in Figure 1 which shows the mean choice of m over all supergames of all sessions of each of our six treatments. Also included is a one-standard error bar and the

mean announced value of m in the two treatments involving pre-game communication. The Figure shows clearly that the mean choice of m in the commitment treatment is indistinguishable from the Ramsey solution m=20, whereas the mean value of m in the other 5 treatments is significantly greater than 20.



Figure 1: Mean *m* Value Across Treatments

**Finding 1:** Inconsistent with Hypothesis 1a, reputation does not serve as a substitute for commitment in any of the five discretionary regimes.

Two-sided Mann-Whitney U-tests show that the money supply under commitment is smaller than in any of the other treatments (p<1%), while there are no significant differences between these other treatments (p>5%). A two-sided Wilcoxon matched pairs test cannot reject that m=20 under commitment (p=15%), but rejects this efficient value for all other treatments (p<1%). The hypothesis that average money supply is equal to the point prediction of the one-period Nash equilibrium (m=60) can be rejected for cheap talk (p=3.5%) and economic transparency (p=1.6%) treatments, but not for the other three discretionary treatments (p>10%). These results suggest that the one-shot Nash equilibrium may be highly relevant in a repeated game, even if all of the theoretical conditions necessary for the Ramsey solution being an

equilibrium hold. Furthermore, neither cheap talk, nor policy transparency, nor economic transparency are effective in reducing the inflation bias, which is consistent with Hypothesis 2a.



Figure 2: Average forecast - m - 20

**Finding 2:** Inconsistent with Hypothesis 3, private sector expectations are biased in the five discretionary treatments, though not in the commitment treatment. In the discretionary treatments, private sector agents systematically under-predict inflation.

Evidence in support of Finding 2 is presented in Figure 2. Since inflation is equal to the money supply plus a transmission shock having an average value of 20, unbiased forecasts should equal m+20. While this hypothesis cannot be rejected for the commitment treatment (p=7.8%), we can clearly reject it for all discretionary treatments (p<1%), and the direction is always negative, indicating under-prediction of inflation. Comparing the different treatments we find that under commitment, the expectation bias is smaller than in all discretionary treatments (p<1%), under cheap talk, the bias is larger than in the other discretionary treatments (p<1.1%). PT, PT+CT, and ET produce similar biases (p>40%) and they are all

larger than under baseline discretion (p<3%). The existence, direction and size of these expectation biases are surprising, and we will discuss some possible explanations below.

An immediate consequence of biased expectations, is that average unemployment is smaller than the NAIRU. Figure 3 shows the average final Container-1 amount that represents unemployment in our model. While the target value is 120, <u>average</u> unemployment in any rational expectations equilibrium (NAIRU) is 140. Since private sector expectations fall short of average inflation rates, the unemployment rate deviates from the natural rate towards the central banks' target rate (except for commitment). Under cheap talk, the effect is so strong that average unemployment is closer to target than to the natural rate.



Figure 3: Average unemployment (final Container 1 amount)

Besides the different inflation biases across treatments and the different abilities of central banks to stabilize employment, the systematic deviations of average unemployment from the natural rate provide a third and unexpected factor influencing central bank welfare. For testing Hypothesis 8, we use two different measures: we compare the actual payoffs of our central bankers as a measure of central bank welfare, but we also compare that payoff with the payoff that the central bank would have achieved in the Ramsey equilibrium had the central banker played an optimal response to shocks. The advantage of the second measure is that it eliminates fluctuations in payoffs stemming from shocks. The first measure, however, allows for a direct test of the point predictions arising from the various equilibria.



Figure 4: Achieved welfare versus equilibrium welfare

**Finding 3:** In most discretionary treatments, average welfare is closer to the one-shot Nash equilibrium and significantly smaller than under commitment. Only cheap talk works as a substitute for commitment regarding welfare.

Figure 4 shows that average welfare is below the welfare level associated with optimal policy in all treatments (p<4%). However, since our subject central bankers are not perfect, they also make mistakes under commitment, so that the payoffs of A-players in this treatment are smaller than predicted by equilibrium (p<1%). If we compare the achieved payoffs between different treatments, we cannot reject Hypothesis 8 for the cheap talk treatment. Here, the achieved payoffs are not smaller than under commitment (p=23%), while they are significantly smaller for the other four discretionary treatments

(p<1.1%). Comparing the payoffs for cheap talk with the other discretionary treatments directly, the evidence is mixed: central bank payoffs are higher under cheap talk than for policy transparency (p=2.1%) and economic transparency (p=4.99%), but not significantly different from baseline discretion (p=8.3%) and PT+CT (p=13%). Comparing welfare with the predictions of the one-period Nash equilibrium, we can clearly reject equality under cheap talk and economic transparency (p=1.6%), but not for the other discretionary treatments (p>80%).

Regarding Hypothesis 9, we can reject that central bank payoffs under economic transparency equal those under commitment (p<1%), but we cannot reject that they equal the payoffs in the baseline discretionary treatment (p=96%). For economic transparency, the one-shot Nash equilibrium and the most efficient equilibrium are both associated with lower payoffs than the corresponding equilibria under baseline discretion. This is because monetary policy cannot stabilize employment in equilibrium if its responses to supply shocks are anticipated. However, the forecasts of our B-players respond less to supply shocks than necessary to offset the responses by money supply, which results in some stabilization of output. In effect the stabilization of output is almost as good as under baseline discretion (see Table 4 below), so we find similar levels of central bank welfare in both treatments.

In order to test hypotheses regarding the players' responses to each other and to supply shocks, we report on treatment-specific, fixed effects regressions where the individual unit (for which fixed effects are allowed) is CB (Type A) subject i of matching group j in supergame k. The main regression for central bank behavior was

$$m_t^i = \alpha + \beta_1 u_t + \beta_2 avgf_t$$

where  $m_t^i$  is the CB player i's money supply choice at time t, and  $u_t$  and  $avgf_t$  are, respectively, the supply shock and average inflation forecast of the private sector that CB player i faced at time t. We impose the additional restriction that  $\beta_2=0$  for the commitment treatment (where the central bank had to decide on m before knowing the private sector's forecast). The regression results are displayed in Table 2.

Treat. Parameter	Commitment	Discretion	Cheap Talk	Policy Transparency	CT + PT	Economic Transparency
α	-35.91***	-2.75	-49.29***	-23.69**	-30.43***	-32.21***
	(5.68)	(7.62)	(5.79)	(9.43)	(8.14)	(7.20)
$eta_1$	0.43***	0.35***	0.50***	0.48***	0.48***	0.40***
	(0.04)	(0.04)	(0.04)	(0.05)	(0.05)	(0.06)
$\beta_2$	-	0.14**	0.49***	0.21**	0.30***	0.46***
		(0.06)	(0.05)	(0.09)	(0.07)	(0.08)
$\mathbb{R}^2$	0.07	0.25	0.57	0.22	0.38	0.38

**Table 2: Central Bank Behavior** 

The estimated coefficient,  $\beta_1$ , is significantly positive in all treatments (p<1%) indicating that central bankers are responding to supply shocks in all six treatments. The significantly positive estimate for  $\beta_1$  in the commitment treatment is at odds with the equilibrium prediction that central bankers should ignore the supply shock altogether (provided that forecasters are rational). The estimates for  $\beta_1$  in the five discretionary treatments are not significantly different from the estimated  $\beta_1$  in the commitment treatment, so that we cannot reject Hypothesis 1b, that in the repeated game, the CB responds to supply shocks as if s/he were operating under a commitment regime. In the five discretionary treatments, the estimates for  $\beta_1$ are significantly less than the equilibrium predictions (from Table 1) of 2/3 (p<5%). We also find that central bankers respond to average inflation forecasts with  $\beta_2$  coefficients that are significantly positive, but also significantly less than the 2/3 prediction of the one-shot Nash equilibrium (p<5%) in all of the discretionary treatments except economic transparency where we cannot reject the null that  $\beta_2 = 2/3$ . Hypothesis 2b, that  $\beta_1 = \beta_2$  in each of the five discretionary regimes, is rejected for the discretionary, policy transparency (PT) and cheap talk plus policy transparency (CT+PT) treatments according to an Ftest (p<5%). In the other two discretionary treatments, cheap talk and economic transparency, CB players are attaching approximately equal weight to supply shocks and inflation expectations, consistent with Hypothesis 2B. We can also make comparisons across discretionary treatments as to whether the coefficient estimates for either  $\beta_1$  or  $\beta_2$  differ pairwise between discretionary treatments. Using a Z-test, we find that  $\beta_1$  is significantly greater in the cheap talk, PT and CT+PT treatments as compared with the discretion treatment (p<5%). Additionally,  $\beta_2$  is significantly greater in the cheap talk and ET treatments as compared with the discretionary and PT treatments and  $\beta_2$  is significantly greater in cheap talk treatment as compared with the CT+PT treatment (p<5%). Thus, it appears that the addition of cheap talk in particular, helps to raise the responsiveness of the central bank to supply shocks and average inflation expectations; indeed, coefficient estimates for  $\beta_1$  and  $\beta_2$  are highest in the cheap talk treatment. Summarizing we have:

**Finding 4:** Central bank responses to supply shocks are positive in all treatments, contrary to the equilibrium prediction for the commitment regime. The estimated weights attached to supply shocks and inflation expectations are highest for the cheap talk treatment but are still lower than equilibrium predictions for all discretionary treatments and the estimates are often significantly unequal, which is inconsistent with Hypotheses 2B.

We next turn to an analysis of the behavior of type B players (private sector agents). We estimate a regression of individual inflation forecasts,  $f_t^i$ , using treatment-specific information available to private sector agents when forming those forecasts including money supply,  $m_t$ , the CB's announcement,  $ann_t$ , the supply shock,  $u_t$ , lagged own expectations, the lagged own forecast error, and the lagged average forecast. While the lagged forecast error can identify eventual learning behavior, the coefficient on the lagged average forecast provides a measure of convergence in forecasts. Specifically, we report in Table 3 on a

$$f_t^i = \alpha + \delta_1 m_t + \delta_2 ann_t + \delta_3 u_t + \beta_1 f_{t-1}^i + \beta_2 \pi_{t-1} + \beta_3 avgf_{t-1},$$

where the individual unit (for which fixed effects are allowed) corresponds to type B subject i of matching group j in supergame (sequence) k. Results are displayed in Table 3.

Treat. Parameter	Commitment	Discretion	Cheap Talk (1)	Cheap Talk (2)	Policy Transparency	CT + PT (1)	CT + PT (2)	Economic Transparency	
α	18.81***	13.97***	36.52***	29.28*** 29.45***		46.29***	43.58***	-20.46***	
	(0.58)	(1.79)	(1.97)	(4.26)	(2.60)	(2.90)	(4.78)	(4.37)	
$\delta_1$	0.96***	-	-	-	-	-	-	-	
	(0.01)								
$\delta_2$	-	-	0.42***	-	-	0.18***	-	-	
			(0.02)			(0.02)			
$\delta_3$	-	-	-	0.08***	-	-	0.05*	0.52***	
				(0.03)			(0.03)	(0.02)	
$\beta_1$	0.01	0.13***	0.04	0.04	0.18***	0.03	0.03	-0.05	
	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	(0.03)	
$\beta_2$	0.05***	0.29***	0.13***	0.11***	0.14***	0.10***	0.08***	0.12***	
	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.02)	(0.02)	(0.02)	
$\beta_3$	-0.03	0.39***	0.09**	0.15***	0.27***	0.17***	0.18***	0.17***	
	(0.03)	(0.03)	(0.04)	(0.05)	(0.04)	(0.05)	(0.05)	(0.05)	
$\mathbb{R}^2$	0.80	0.74	0.38	0.46	0.58	0.41	0.52	0.25	

**Table 3: Private Sector Forecasts** 

Note: \*\*\* significant at 1%, \*\* significant at 5%, \* significant at 10%, numbers in parantheses are estimated standard deviations.

In the Commitment treatment, theory predicts that  $\delta_1 = 1$ , and while we can reject this prediction (p<5%), the actual estimate (.96) is very close to 1. Under economic transparency, consistent with Hypothesis 5, subjects respond to supply shocks, but the coefficient,  $\delta_3$ , is significantly smaller than the theoretical prediction of 2 (p<1%). Note that for both cheap-talk treatments, the response of inflation forecasts to announcements made by the central bank is positive and significant as indicated by the estimate for  $\delta_2$ . If cheap talk is combined with policy transparency, the announcement coefficient is reduced by more than 50%. This shows that our subject central bankers are successful in affecting expectations by non-binding announcements, in particular when these announcements cannot be fully evaluated by forecasters against actual decisions by the central bankers as in the cheap talk treatment. This finding may come as a surprise and it explains why we observe the largest bias between actual and expected inflation in the cheap-talk treatment without policy transparency. We will further analyse CB announcements below, to determine whether central bankers try to exploit their effect on expectations.

In a second specification for the two cheap talk treatments (2) as reported in Table 3, we also explore whether unobserved supply shocks affect individual forecasts via CB announcements. We do this by setting  $\delta_2 = 0$  and adding the unobserved (in these treatments) supply shock,  $u_t$ . We find that in both cheap talk treatments, the supply shock does affect forecasts, as evidenced by the positive and significant coefficient estimate for  $\delta_3$  suggesting that the CB *is* communicating this information via its announcements, contrary to Hypothesis 5. By contrast in the economic transparency treatment, private sector forecasts *should* condition on the supply shock with weight 2 (see Table 1) but the estimated weight reported in Table 3 is only about one quarter of this prediction.

Summarizing we have:

**Finding 5:** Consistent with theory, under commitment, forecasts are closely aligned with the monetary policy choice. Inconsistent with theory, in the cheap talk treatments, forecasts respond to policy announcements and, these announcements reveal information about supply shocks. The response of inflation forecasts to supply shocks in the economic transparency treatment is about 25% of the predicted level.

Table 3 further reveals that in all treatments, subjects' forecasts respond to lagged inflation as evidenced by the significantly positive coefficient estimates for  $\beta_2$ , which suggests evidence of adaptive learning. Numerically, this effect is smallest for commitment and highest for the pure discretionary treatment. Further, as  $\beta_3$  is significantly positive in all discretionary treatments, individual forecasts converge over time.<sup>9</sup>

<sup>&</sup>lt;sup>9</sup> Lagged own and average forecasts are highly correlated, in particular under commitment. We used a 2-step GLS estimate as a robustness check, and the qualitative results were the same.

If forecasters are learning, it is puzzling that their expectations are systematically biased towards too low an inflation rate, which runs counter to Hypotheses 3 and 4. Nevertheless, it may be that this bias becomes smaller over time. In order to address that question, we analyze the data on expectation bias separately for the first and last 15 periods of each treatment<sup>10</sup> and focus on the magnitude of the bias in expectations and on the welfare effects that are heavily influenced by average unemployment deviating from NAIRU for biased expectations. This analysis is presented in Figures 5 (expectation bias) and 6 (welfare). We find that the expectation bias is indeed smaller for the later periods than for the first periods in all discretionary treatments (see Figure 5), but this difference is significant only for the baseline discretionary treatment and for the PT+CT treatment (p<2%; in other treatments: p>10%). In the CT treatment (and only there), the average money supply is significantly higher in the later periods than in the early periods (p<1%).<sup>11</sup> Rising average inflation and unemployment reduce the achieved welfare levels in the CT treatment (see Figure 6): the average welfare level in the last 15 periods is at 72.8% of the first best. This is smaller than under commitment (p=8.3%) and the differences between CT and the other four discretionary treatments or the one-shot Nash equilibrium are not significant anymore (p>19%). For CT, the difference is highly significant (p<1%). For all other treatments, the differences are not significant (p>19%). Summarizing, our main result from examining expectations and welfare over time is:

**Finding 6:** The expectation bias in favor of low inflation diminishes with experience in all discretionary treatments. Cheap talk initially raises welfare to levels approximating the first best, but this effect is only temporary (first 15 periods).

<sup>&</sup>lt;sup>10</sup> In sessions combining policy transparency with cheap talk, we had only 25 to 32 periods in total due to the extra input and feedback required by this treatment. Thus, for the CT+PT treatment sessions, we took the first and last 12 periods instead. The other sessions had 30 to 49 periods.

<sup>&</sup>lt;sup>11</sup> We observe a significant reduction in money supply under commitment (p=5.5%). In other treatments, there are no significant differences between money supply in the first and last periods (p>19%).



Note: In CT+PT we compared the averages from the first and last 12 periods.



Figure 6: Welfare in the first and last 15 periods

Note: In CT+PT we compared the averages from the first and last 12 periods.

Next, we look at the stability of unemployment across treatments. Due to the combined responses of monetary policy, announcements, and expectations (when it applies) to supply shocks, there may be significant differences in the stability of unemployment across treatments. Just looking at the standard deviations of unemployment, we find that it is significantly larger than predicted by theory in all treatments. In fact, our subject central bankers even contribute to fluctuations in unemployment, because standard deviations are usually higher than they would be for a constant money supply. Table 4 displays the standard deviation of unemployment, averaged over all matching groups for each treatment along with equilibrium predictions, repeated here from Table 1.

	Commitment	Discretion	Cheap Talk	Policy Transparency	CT + PT	Economic Transparency
$std(y_t)$	16.56	19.48	17.73	18.66	19.03	20.55
equilibrium	16.33	12.17	12.17	12.17	12.17	16.33

**Table 4: Standard Deviation of Unemployment** 

Two-sided Wilcoxon matched pairs test show that fluctuations in unemployment are significantly larger than the equilibrium predictions under the baseline discretion treatment (p=2.3%), economic transparency (p<1%), and the combination of cheap talk and policy transparency (p=3.9%). In the other treatments, there is no significant difference. The reason for the high variation of unemployment in spite of central bankers responding to supply shocks in the right direction are (i) central bankers are changing over time and different central bankers are pursuing different policies, (ii) forecasters and their forecasts are changing over time and fluctuating expectations contribute to fluctuations in employment, even if central bankers respond to expectations in an optimal way,<sup>12</sup> and (iii) as we have found, central bankers respond with suboptimal coefficients to supply shocks and expectations, so that the combination of both may leave unemployment with more fluctuations as under constant expectations and constant money supply. With specific regard to Hypothesis 6, we have the following:

<sup>&</sup>lt;sup>12</sup> Note that expectations are constant in equilibrium (except for economic transparency) but fluctuate quite a lot in the experiment. An increase in expected inflation has the same effect on the economy as an adverse supply shock.

# **Finding 7:** Contrary to Hypothesis 6, unemployment (output) volatility is not generally lower in the commitment and economic transparency treatments as compared with the other discretionary treatments.

Support for Finding 7 comes from two-sided Mann-Whitney-U tests which indicate that unemployment fluctuates less under commitment than under the baseline discretion (p=2.1%) and economic transparency (p<1%) treatments. Cheap talk also leads to lower fluctuations than economic transparency (p=2.1%). The other pairwise comparisons do not yield significant differences (p>10%). It is remarkable that discretionary monetary policy leads to higher fluctuations of employment than commitment (rejecting Hypothesis 6). The reason is that in the commitment regime, inflation expectations and the money supply move almost 1 to 1 (see the coefficient  $\delta_1$  in Table 3). Hence, the only remaining impact on employment comes from exogenous shocks. In the discretionary treatments, changing CB policies and fluctuations of expectations that are not well coordinated with monetary policy lead to higher and unsystematic fluctuations between actual and expected inflation and work to destabilize employment. This result was also surprising to us. It provides a strong argument in favor of rule-based monetary policy. The fact that cheap talk leads to the second lowest standard deviation of employment indicates that announcements could have a small stabilizing effect. Of course, this requires that actual policy is correlated with announcements, as is indeed the case (as we will show below). However, since the differences in std(y) between cheap talk and most other treatments are not significant, we cannot draw any conclusions for the eventual stabilizing effects of cheap talk. This topic requires further exploration.<sup>13</sup>

	Commitment	Discretion	Cheap Talk	Policy Transparency	CT + PT	Economic Transparency
St.Dev. of $\pi_t$	18.86	19.94	19.60	18.44	17.93	20.28
Equilibrium	11.55	13.88	13.88	13.88	13.88	25.82

**Table 5: Standard Deviation of Inflation** 

<sup>&</sup>lt;sup>13</sup> Ahrens et al. (2016) are currently working on a comparable experiment in which they focus on the effects of forward guidance.

Volatility of inflation is about the same across all treatments (see Table 5). In the treatment CT+PT, it is significantly smaller than for the baseline discretionary treatment (p=4.0%) or under economic transparency (p=1.4%). PT also leads to smaller inflation volatility than ET (p=5.2%). All other pairwise comparisons are insignificant (p>0.17). With the sole exception of the ET treatment, inflation volatility is significantly greater than in rational expectations equilibrium (p<1%); under economic transparency, inflation volatility is smaller than predicted (p<1%). The high level of inflation volatility can be explained once more by central bankers pursuing different policies over time. Under economic transparency, inflation volatility is predicted to be higher, because in equilibrium, private sector expectations respond to the shock so strongly that the CB has no incentive to deviate. The previous regressions, however, have shown that inflation expectations and monetary policy respond to supply shocks with coefficients that are much smaller than in equilibrium and the monetary policy response under economic transparency is rather conparable to the other discretionary treatments. For this reason, the volatility of inflation under economic transparency is not significantly higher than in those other treatments and, thus, lower than predicted by equilibrium theory. In the literature, it has been argued that imprecise public announcements may raise the volatility of aggregate prices (Amato and Shin, 2003). The evidence in our experiment does not support this view, but the structure of our model differs from Amato and Shin in that our forecasters have no incentives to coordinate their forecasts.

The dispersion of individual forecasts within a group of forecasters provides a good measure for the overall transparency of the policy regime. If CB behaviour is fairly predictable, then individual forecasts should be close to one another. In equilibrium, all B-players should have the same expectations leading to a zero dispersion of individual forecasts independent from the treatment. We find, however, contrary to Hypothesis 4, that there are significant differences in this dispersion across treatments (see Table 6).

	Commitment	Discretion	Cheap Talk	Policy Transparency	CT + PT	Economic Transparency
St.dev. $(\pi_t^i t)$	5.56	8.94	11.57	9.16	11.09	12.28
Avg. Payoff of B-players	3799	3537	3426	3515	3475	3451

 Table 6: Average Dispersion of Forecasts Within Groups and Forecasters' payoffs

In the commitment treatment, the dispersion of individual forecasts is smaller than in all discretionary treatments (p<1%). Within discretionary treatments, the dispersion is higher when forecasters get *more* information, as under cheap talk, CT+PT, or economic transparency. The differences between dispersion in these treatments vis-á-vis the baseline discretionary or policy transparency treatment are all significant (p<10%), while differences within this group or between baseline and policy transparency are not significant (p>10%). Without these signals, private sector agents can more easily coordinate on a common expectation. These findings contrast with empirical results by Ehrmann et al. (2012), who "find that central bank transparency and communication (...) reduce the dispersion of professional forecasters' views" (p.1019). Hubert (2015), however, attributes the impact of central bank communication on professional forecasts to their policy dimension rather than their economic accuracy.

Since our B-players are paid according to forecast accuracy, their payoffs measure the quality of their forecasts. Under commitment, these payoffs are greater than in all discretionary treatments (p<1%). We find no significant differences in B-players' payoffs between discretionary treatments (p>10%), except that payoffs under economic transparency are somewhat smaller relative to baseline discretion (p=7%). Summarizing, we have:

# **Finding 8:** Inconsistent with Hypothesis 4, distribution of individual forecasts around the average inflation forecast differs across treatments.

Why don't B-type subjects use cheap-talk announcements or information about supply shocks to better coordinate their expectations than in the baseline treatment without this information? The data indicate that subjects disagree in how actual money supply is related to announcements or supply shocks. Different perceptions of CB credibility create a huge dispersion in forecasts. Further, under economic transparency, subjects can have different expectations about the CB's responses to supply shocks. If these expectations diverge, the posterior beliefs after learning the supply shock may be more dispersed than prior beliefs

without such information. While in theory, economic transparency raises forecasters' payoffs compared to baseline discretion, we observe just the opposite in our experimental data.

As cheap talk seems to have a remarkable impact on the economies, we also analyse how the cheaptalk messages depend on various explanatory variables. Here, we use three different specifications which are versions of the regression model:

$$ann_{t}^{i} = \alpha + \delta_{1} u_{t} + \beta_{1} ann_{t-1}^{i} + \beta_{2} (avgf_{t-1} - ann_{t-1}^{i}) + \beta_{3} m_{t-1}^{i} + \beta_{4} \pi_{t-1}$$

where the variable names are as described earlier. Table 7 reports on fixed effects regression results where the individual unit is as described previously for Table 2.

Treat.	Cheap Talk	Cheap Talk	Cheap Talk	CT + PT	CT + PT	CT + PT
Parameter	(1)	(2)	(3)	(1)	(2)	(3)
α	-26.43***	-28.72***	-23.54***	2.45	18.75	12.46
	(8.38)	(8.79)	(8.57)	(10.64)	(12.16)	(12.01)
$\delta_1$	0.28***	0.28***	0.28***	0.19***	0.17**	0.20***
	(0.06)	(0.06)	(0.06)	(0.07)	(0.08)	(0.08)
$eta_1$	0.02	-	-	0.04	-	-
	(0.06)			(0.04)		
$\beta_2$	-	0.05	0.04	-	-0.20***	-0.20***
		(0.06)	(0.06)		(0.06)	(0.06)
$\beta_3$	0.05	0.06	-	-0.19	-0.26***	-
	(0.06)	(0.06)		(0.06)	(0.07)	
$eta_4$	-	-	-0.02	-	-	-0.15***
			(0.04)			(0.05)
R <sup>2</sup>	0.08	0.02	0.01	0.04	0.08	0.11

**Table 7: Cheap Talk Announcements** 

Note: \*\*\* significant at 1%, \*\* significant at 5%, numbers in parantheses are estimated standard deviations.

Note first, and most importantly, that for both cheap talk treatments, and all regression specifications, the CB's announcements vary positively and significantly with the supply shocks as evidenced by the significantly positive estimates for  $\delta_1$ . Thus, these CB announcements are informative about the current

state of fundamentals, and private sector forecasters can extract some information from them. However, given the low R<sup>2</sup> and the low coefficient estimates by which CB-players respond to these supply shocks, the forecasters should put a very low weight on these announcements, which is indeed in line with the low weights given to CB announcements as reported in the regressions reported in Table 3. Notice second that there is no persistence in announcements, as evidenced by the insignificant coefficient attached to  $\beta_1$ . Finally, the significantly negative coefficient estimates for  $\beta_2$  in the CT+PT treatment suggest that the greater the departure of lagged inflation forecasts from the lagged announcements, the more the CB attempts to dampen down inflationary pressure by announcing an even lower policy choice for  $m_t$ . Eventually, it seems, the private sector learns to ignore these cheap talk messages because of the evident disconnect between the CB announcements and actual inflation as revealed in Figure 1, but this does not prevent the CB from trying to use cheap talk announcements to influence private sector expectations. Summarizing, we have:

**Finding 9:** Cheap talk announcements by CB players are informative as they convey information about supply shocks. When there is also policy transparency, CB players react to departures of inflation forecasts from lagged announcements by lowering their announced policy for  $m_t$ .

### 6. Conclusion

Central bankers operate in a discretionary world, where they face a trade-off between credibility in stabilizing inflationary expectations on the one hand and flexibility in response to economic shocks on the other. In this paper we have posed the question of whether a balance can be found between these twin objectives in a repeated game setting, where reputational concerns might serve as a substitute for commitment so that welfare under discretionary policy might approximate or even exceed the levels attainable under a full commitment regime. We addressed this question using a version of the Barro-Gordon (1983ab) monetary policy game model and controlled laboratory experiments with paid human subjects serving in the role of private sector agents forecasting inflation or as central bankers facing the

policy trade-off between credibility and flexibility. The advantage of our laboratory approach is that we have good control over the incentive structure and information available to all actors so that we can formulate crisp equilibrium predictions about the type of behavior we should observe under commitment and discretionary regimes. This same level of control is not available using field data so that an evaluation of whether central banks are behaving in a discretionary repeated game setting as if they had a commitment is not really testable in the field.

In answer to the question we posed, we can unambiguously reject the notion that reputation in the discretionary repeated game setting leads to outcomes approximating the first best, Ramsey solution in favor of the alternative that behavior is closer to the predictions of the one-shot Nash equilibrium, involving higher inflation and lower welfare. Importantly, as we also study the full commitment regime in the laboratory, we are able to show that there are important policy and welfare differences between the commitment and discretionary regimes. In addition, we have considered several augmented versions of our baseline discretionary policy regime that allow central bank cheap talk about policy intentions, ex-post transparency about policy actions, both cheap talk and policy transparency and economic transparency. Among these mechanisms, only cheap talk served to raise welfare to levels approximating those achieved under the commitment regime, as central bankers tried to convince private sector agents that they would keep inflation low. However even this effect diminished over time, as private sector agents learned to ignore the CB's cheap talk messages. We conclude that the discretionary regimes of our laboratory study are indeed welfare reducing relative to the commitment regime, and that flexibility in responding to economic shocks and expectations appears to dominate any longer-term concerns about credible maintenance of low inflationary expectations. This welfare conclusion is further supported by the observation that private sector forecasters earn the highest payoffs under a commitment regime.

It is remarkable that we do not find a trade-off between credibility and flexibility. Indeed, we find that discretionary policy regimes which allow for the stabilization of employment actually lead to *higher* variations in employment than under a perfectly credible commitment regime where stabilization is not

possible, although central bankers do make an attempt to stabilize the impact of supply shocks in all regimes. As we have seen, there are several reasons for this divergence between theory and outcomes. First, central-bank responses to shocks are smaller than optimal in all of our discretionary treatments, diminishing the stabilizing role of those more flexible discretionary regimes. Second, changes in strategies by different central bankers provide an additional source of strategic uncertainty that is not addressed by the theory. As this strategic uncertainty is likely to carry over to the field, we conclude that central banks would do well to follow strict rules and make them common knowledge among forecasters. Any changes in central bank strategies should be communicated thoroughly and before they are implemented, to minimize strategic uncertainty. Third, forecasters are unable to effectively use the additional information provided by central bank announcements or about shocks (under economic transparency) to improve their forecasts. The reason is that forecasters disagree about the quality of announcements and the likely response of central bankers to supply shocks. Unlike all theoretical predictions, this strategic uncertainty raises forecast errors relative to the baseline discretionary regime in which forecasters do not get any additional information regarding the realization of the current period's random variables. From the latter result we conclude that CBs should avoid providing ambiguous signals or information about fundamentals that leave the private sector puzzled about the likely response of inflation to those fundamentals.

We expect that our experimental findings carry over to the "real world" as well since the incentives and uncertainties that our subjects faced also approximate those faced by real central bankers and private sector agents. For instance, we note that there is corroborating empirical evidence that central banks publish inflation forecasts to strategically manipulate private inflation forecasts just as our subjects central bankers do in the treatments with cheap talk (Gomez-Barrero and Parra-Polania, 2014). While it is possible that well-intentioned, real world central bankers, aware of the time inconsistency problem of monetary policy, can learn to implement the optimal policy – in the words of McCallum (1995), they "just need to do it" – the long debate about rules versus discretion in central bank policy suggests that there are also doubts about the ability of such real world central bankers to effectively manage the trade-off between credibility and

flexibility. The experimental evidence that we have presented in this paper provides further evidence that such doubts may be warranted.

## References

Ahrens, S., J. Lustenhouwer, and M. Tettamanzi (2016), "The Stabilizing Role of Forward Guidance: A Macro Experiment," mimeo.

Amato, J.D., and H.S. Shin (2003) "Public and Private Information in Monetary Policy Models," BIS Working Papers No. 138.

Arifovic, J. and T.J Sargent (2003) "Laboratory Experiments with an Expectational Phillips Curve," in D.E. Altig and B.D. Smith (eds.) Evolution and Procedures in Central Banking, Cambridge University Press.

Arifovic, J., H. Dawid, C. Deissenberg, and O. Kostyshyna, (2010) "Learning Benevolent Leadership in a Heterogenous Agents Economy," Journal of Economic Dynamics and Control 34, 1768-1790.

Arifovic, J. (2014) "Evolving Better Strategies for Central Bank Communication: Evidence from the Laboratory," Experiments in Macroeconomics: Research in Experimental Economics Vol. 17, 229-258.

Barro, R.J., and D.B. Gordon (1983a) "A Positive Theory of Monetary Policy in a Natural Rate Model," Journal of Political Economy 91(4), 589-610.

Barro, R.J., and D.B. Gordon (1983b) "Rules, Discretion and Reputation in a Model of Monetary Policy," Journal of Monetary Economics 12, 101-121.

Blinder, A.S. (1999) Central Banking in Theory and Practice. Cambridge, MA: MIT Press.

Camera, G., M. Casari, and M. Bigoni (2011) "Communication, Commitment, and Deception in Social Dilemmas: Experimental Evidence," University of Bologna working paper DSE No. 751.

Crawford, V. (1998) "A Survey of Experiments on Communication via Cheap Talk," Journal of Economic Theory 78, 286-298.

Duffy, J. and N. Feltovich (2006), "Words, Deeds and Lies: Strategic Behaviour in Games with Multiple Signals," Review of Economic Studies 73, 669–688

Ehrmann, M., S. Eijffinger, and M. Fratzscher (2012) "The Role of Central Bank Transparency for Guiding Private Sector Forecasts," Scandinavian Journal of Economics 114, 1018–52.

Engle-Warnick, J., and N. Turdaliev (2010) "An Experimental Test of Taylor-type Rules with Inexperienced Central Bankers," Experimental Economics 13, 146-166.

Fischbacher, U. (2007) "z-Tree: Zurich Toolbox for Ready-made Economic Experiments," Experimental Economics 10(2): 171-178.

Frechette, G.R. (2015) "Laboratory Experiments: Professionals Versus Students" in G.R. Frechette and A. Schotter, (Eds.) Handbook of Experimental Economic Methodology, Oxford University Press.

Geraats, P. (2002) "Central Bank Transparency," Economic Journal 112, F532-F565.

Geraats, P. (2014) "Monetary Policy Transparency," Chapter 3 in: J. Forssbæck and L. Oxelheim (Eds.), The Oxford Handbook of Economic and Institutional Transparency, Oxford University Press.

Gomez-Barrero, S. and J.A. Parra-Polania (2014) "Central Bank Strategic Forecasting," Contemporary Economic Policy 32, 802-810.

Henckel, Timo, Gordon D. Menzies, Nick Prokhovnik, and Daniel J. Zizzo (2011) "Barro-Gordon Revisited: Reputational Equilibria with Inferential Expectations," Economics Letters 112, 144-147.

Hubert, P. (2015) "Do Central Bank Forecasts Influence Private Agents? Forecasting Performance Versus Signals," Journal of Money, Credit, and Banking 47, 771–89.

Kydland, F., and E.C. Prescott (1977) "Rules Rather than Discretion: the Inconsistency of Optimal Plans," Journal of Political Economy 85, 473–491.

Lucas, R.E., Jr. (1986) "Adaptive Behavior and Economic Theory," Journal of Business 59, S401-S426.

McCallum, B. (1995) "Two Fallacies Concerning Central-Bank Independence," American Economic Review 85, 207-211.

Taylor, J.B. (1999) "A Historical Analysis of Monetary Policy Rules," in: J.B. Taylor, (Ed.) Monetary Policy Rules. Chicago: University of Chicago Press, pp. 319 – 348.

Van Huyck, J.B., R. Battalio and M.F. Walters (1995) "Commitment versus Discretion in the Peasant-Dictator Game," Games and Economic Behavior 10(1), 143-171.

Van Huyck, J.B., R. Battalio and M.F. Walters (2001) "Is Reputation a Substitute for Commitment in the Peasant-Dictator Game?" working paper, Texas A&M University.

Walsh C.E. (1995) "Optimal Contracts for Central Bankers" American Economic Review 85, 150-167.

# Appendix: Instructions Used in the Baseline Discretionary Treatment (Other Instructions are Similar)

# 1. Overview

Welcome to this experiment in the economics of decision-making. Please read these instructions carefully as they explain how you earn money from the decisions you make in today's experiment. There is no talking for the duration of today's session.

We will first go over these instructions. When we are done, each of you will have to answer a few brief questions to ensure that all participants understand these instructions. You will also have time to ask clarifying questions. Then, you will begin making your decisions using the computer workstations.

Today's session consists of a number of "sequences". Each sequence consists of a number of "rounds". At the start of each sequence the computer program will randomly assign all participants to a 5-member group. All random groupings of 5 participants are equally likely. Once you are assigned to a 5-member group, you will play all rounds of the sequence with the same 4 other members of your group. At the start of each new sequence, the computer program will again randomly assign players to 5-member groups. Your interactions with other participants is always anonymous; you will not be informed of the identity of any group member in any sequence played, nor will they be informed of your identity, even after today's session is over.

Prior to the first round of each new sequence, the program randomly selects one member of your 5member group and assigns that person the role of Player A. The other 4 members of your group are assigned the role of Player B. You and the other 4 participants will remain in your role of Player A or B for all rounds of the sequence. At the start of each new sequence, the computer program will once again assign roles randomly among the members of your new 5-member group, and you will remain in your new role for the duration of that new sequence.

# 2. The decisions to be made

Imagine there are two containers labeled Container 1 and Container 2. At the start of each round, Container 1 holds  $W_0$  gallons of water while Container 2 is empty.

In each round, the four Player Bs in each group move first. Each Player B submits his forecast as to how many gallons of water there will be in Container 2 at the *end* of the round.

After all Player Bs have made their forecasts, the computer program calculates the average of the four Player B forecasts, which we denote by af. This average forecast is added to the amount of water in Container 1 so that the total amount of water in Container 1 is now  $W_0 + af$ .

Next, Player A learns both  $W_0$  and af and thereby the total amount of water in Container 1. Then, Player A can move up to 80 gallons of water from Container 1 to Container 2. Denote the amount moved by M.

In addition, there is an uncontrolled flow of water V from Container 1 to Container 2. Thus, the final amount of water in Container 2 is M + V.

# 2.1. Specific details

The initial water level in Container 1,  $W_0$ , is a random variable. For each round of a sequence, the computer program draws the value of  $W_0$  randomly from a **uniform distribution over the interval [120, 160]**, which means that the minimum possible value of  $W_0$  is 120 and the maximum possible value of  $W_0$  is 160. All numbers between 120 and 160 inclusive have an equal chance of being drawn, so the expected value of  $W_0$  is 140.

In each round, the four **Player B**'s in each group move first. Each must submit a forecast, f, of the amount of water that will be in Container 2 at the end of the round. Recall that Container 2 is initially empty. Forecasts may range from 0 to 120 units of water inclusive in Container 2. Player Bs should type their forecast in the input box on the first decision screen when prompted. Click the OK button when satisfied with your choice.

After all four Player Bs have entered their forecasts, the computer program calculates the *average value of the four forecasts*. We denote this average forecast by *af*. Then, *af* units of water are added to Container 1. So, the average forecast increases the amount of water in Container 1.

The total amount of water in Container 1 is now  $W_{\theta} + af$ .

Note that Player Bs do not know precisely the value of  $W_0$  nor do they know **af**; they do know that  $W_0$  is a uniform draw from [120, 160] and their own forecast, f.

Next, Player A alone is informed of the value of af for the round. In addition, Player A learns this round's value of  $W_0$  and is told the amount of water in Container 1,  $W_0+af$ .

After observing the values of af and  $W_{\theta}$  and the total amount of water in Container 1, the Player A in each group must decide how much water to move from Container 1 to the empty Container 2. Player A can move up to 80 gallons of water inclusive from Container 1 to Container 2 in each round. Let us denote by M the amount of water *moved* by Player A from Container 1 to Container 2. Player A should type his or her choice for M in the input box on the first decision screen when prompted. Click the OK button when satisfied with your choice.

In addition, there is an uncontrolled flow of water from Container 1 to Container 2. This flow is another random variable, denoted by V. The computer program draws the value of V randomly from a uniform distribution over the interval [0, 40], which means that the minimum possible value of V is 0 and the maximum possible value of V is 40. All numbers between 0 and 40 inclusive have an equal chance of being drawn, so the expected value of V is 20. Player A does not know V when deciding how much water to move, M. The uncontrolled flow, V, is determined only after Player A's choice of M has been made.

The final amount of water in Container 1 is:  $W_0 + af - M - V$ .

The final amount of water in Container 2 is M + V.

Participants' payoffs depend on the final amounts of water in Containers 1 and 2 as described in the next section.

# 2.2. Payoffs for the round

If you are a Player A, the final amounts of water in both Containers 1 and 2 are used to determine your payoff in points for the round according to the formula:

Player A Points = 4000 - 2 (Final Container 1 amount - 120)<sup>2</sup> – (Final Container 2 amount - 40)<sup>2</sup>

For your convenience, a non-exhaustive table of values for Player A's payoff in points is given in Table A as a function of the final water levels in Containers 1 and 2. Notice that Player A's maximize their payoff if the final amount of water in Containers 1 and 2 are as close as possible to 120 and 40, respectively, and that deviations in the final Container 1 water amount from 120 are 2 times more costly than are deviations of the final Container 2 water amount from 40.

If you are a Player B, only the final amount of water in Container 2 matters for your payoff in points. Specifically, your payoff in points for the round is given by the formula:

Player B Points =  $4000 - (\text{Final Container 2 amount} - f)^2$ 

Note that f denotes a Player B's *own* forecast for the round (and *not* the average forecast, af). For your convenience, a non-exhaustive table of values for Player B's payoff in points is given in Table B (attached) as a function of the forecast error, Final Container 2 amount -f. Notice that Player B's maximize their payoffs when the forecast error is 0, that is when f=Final Container 2 amount.

# Feedback and record keeping at the end of each round.

At the end of each round, Player As will be reminded of  $W_0$ , *af* and their choice of *M*. They will then be told the value of the random variable *V*, and the final amounts of water in Containers 1 and 2. Finally, Player A's will be told their payoff in points for the round and their cumulative point total for the sequence.

At the end of each round, Player Bs will be reminded of their choice of f. They will then be told the final amounts of water in Containers 1 and 2 and the value of the random variable  $W_{\theta}$ . Finally, they will be told their payoff in points for the round and their cumulative point total for the sequence.

Following revelation of this information, the round is over. Please record the results of the round on your record sheets under the appropriate headings. When you are done recording this information press the continue button. The sequence may or may not continue with a new round. If a new round is played, the procedure will be the same as described above. Following the first round of a sequence, all players will see at the bottom of their screens, a history of past final amounts of water in Containers 1 and 2 for the five-person group they were in along with their own payoff in points for each round and their cumulative payoff in points from all rounds played in a given sequence.

# 3. When does a sequence of rounds continue and when does it end?

At the end of each round, the computer program will randomly draw a number (an integer) between 1 and 6, inclusive. All numbers, 1,2,3,4, 5 and 6 have an equal chance of being drawn; it is like rolling a six-sided die. The number drawn will be displayed on your computer screen. If the number chosen is 1,2,3,4 or 5, the sequence will continue with a new round. If a 6 is chosen, the sequence will end. Thus, there is a 5 in 6 (83.33 percent) chance that a sequence will continue from one round to the next and a 1 in 6 (16.67 percent) chance that the current round will be the last round of the sequence.

Depending on the time available, a new sequence may then begin. At the start of this new sequence you would be randomly formed into new 5-member groups. One member of each group would be randomly chosen to play the role of Player A. The other four members would be assigned the role of Player B. These roles would again remain fixed for the duration of the new sequence.

If, by chance, the final sequence has not ended by the three-hour time period for which you have been recruited, we will schedule a continuation of that final sequence for another time in which everyone here can attend. You would be paid based on your cumulative point total for one randomly selected sequence that finished in today's session and if you had to return for a continuation of a sequence, you would also be paid based on your cumulative point total for that final sequence, once it had finally ended.

# 4. Earnings

If today's session ends within the 3-hour time period for which you have been recruited, then your payoff will depend on the total number of points you earned in a maximum of two of the sequences that were played in today's session. Specifically, if only one sequence was played, then your point total for the session will equal your point total from that sequence. If two or more sequences have been played, then your point total for the session will be the sum of your cumulative point totals from two sequences. One of these two sequences will be the one in which you earned the *highest* cumulative point total from among all sequences played in today's session. The other sequence will be one that is randomly chosen from among all sequences played in today's session. Your session point total from the chosen sequence(s) will be converted into dollars at the rate of 2000 points =\$1.00 (or 20 points = 1 cent). Clearly, the more points you earn the higher is your dollar payoff. Since you don't know in advance which sequence(s) will determine your payoff, you will want to do your best in every sequence. If as mentioned above, the final sequence does not end within the 3 hour time period, then you would be paid for one randomly chosen sequence that ended in today's session (provided that event occurred) and/or for the final sequence, when that final sequence finally ended, in a later continuation session.

In addition to your dollar earnings from the two sequences chosen for payment, you begin each sequence with 5000 points (\$2.50). The 5,000 initial endowment of points will show up in your cumulative point total for each sequence. Since we will pick two sequences for payment, these two initial point balances of 5,000 points (10,000 total) comprise your \$5.00 payment for participation in this session. If only one sequence is played in today's session then we will add another 5000 points to your cumulative point total for that one sequence. Note that your initial or cumulative point total in each sequence is reduced by negative points earned in any round.

# 5. Questions

Now is the time for questions. If you have a question about any aspect of these instructions, please raise your hand and an experimenter will come to you and answer your question in private.

# 6. Quiz

Before the start of the experiment we ask you to answer the following quiz questions in the spaces provided. The numbers in these quiz questions are merely illustrative; the actual numbers in the session

may be quite different. Feel free to consult the instructions and tables to answer these questions. After all participants have completed this quiz we will come around to check your answers.

- Suppose Player A observes that *W*<sub>θ</sub> =130 and *af*=60 so that the new level of water in Container 1 is: 190. Player A then chooses *M*= 70. Suppose it turns out that *V*=25. What is the final amount of water in Container 2 in this case? \_\_\_\_\_ What is the final amount of water in Container 1? \_\_\_\_\_ What is Player A's payoff in points for the round? \_\_\_\_\_ If a Player B forecast f=75, what would be that individual Player B's payoff for the round? \_\_\_\_\_\_
- 2. Same situation as in question 1, except that Player A chooses *M*=40 instead of *M*=70. What is the final amount of water in Container 2 in this case? \_\_\_\_\_ What is the final amount of water in Container 1? \_\_\_\_\_ What is Player A's payoff in points for the round? \_\_\_\_\_ If a Player B forecast f=75, what would be that individual Player B's payoff for the round? \_\_\_\_\_\_
- 3. Suppose Player A observes that W<sub>0</sub>.=150 and af=30 so the new level of water in Container 1 is:180. Player A then chooses M=30. Suppose it turns out that V=15. What is the final amount of water in Container 2 in this case? \_\_\_\_\_ What is the final amount of water in Container 1?\_\_\_\_\_ What is Player A's payoff in points for the round? \_\_\_\_\_ If a Player B forecast f=35, what would be that individual Player B's payoff for the round? \_\_\_\_\_\_
- 4. Same situation as in question 3, except that Player A chooses *M*=10 instead of *M*=40. What is the final amount of water in Container 2 in this case? \_\_\_\_\_ What is the final amount of water in Container 1? \_\_\_\_\_ What is Player A's payoff in points for the round? \_\_\_\_\_ If a Player B forecast f=35, what would be that individual Player B's payoff for the round? \_\_\_\_\_\_
- 5. Suppose it is round 2 of a sequence. What is the chance that the sequence will continue with round 3? \_\_\_\_\_\_. Would your answer change if we replaced round 2 with round 12 and round 3 with round 13? Circle one: yes / no.
- 6. True or false? You will remain in the same role as a Player A or Player B in all rounds of all sequences? Circle one: True / False.
- True or false? You will be paid based on the points you earned in a maximum of two sequences. Circle one: True / False.

#### Final Amount in Container 2 → 20 25 30 35 40 50 55 65 70 80 85 100 105 110 115 Final Amount in 10 11 44 75 95 120 Container 1↓ -12000 -11625 -11300 -11025 -10800 -10625 -10500 -10425 -10400 -10425 -10500 -10625 -10800 -11025 -11300 -11625 -12000 -12425 -12900 -13425 -14000 -14625 -15300 -16025 -16800 10 -9700 -9325 -9000 -8725 -8500 -8325 -8200 -8125 -8100 -8125 -8200 -8325 -8500 -8725 -9000 -9325 -9700 -10125 -10600 -11125 -11700 -12325 -13000 -13725 -14500 -7600 -7600 -7225 -6900 -6625 -6400 -6225 -6100 -6025 -6000 -6025 -6100 -6225 -6400 -6625 -6900 -7225 -8025 -8500 -9025 -9600 -10225 -10900 -11625 -12400 20 -4100 -5700 -5325 -5000 -472 -4500 -4325 -4200 -4125 -412 -4200 -4325 -4500 -4725 -5000 -5325 -5700 -6125 -6600 -7125 -7700 -8325 -10500 30 -9000 -9725 -4000 -3625 -3300 -3025 -2800 -2625 -2500 -2425 -2400 -2425 -2500 -2625 -2800 -3025 -3300 -3625 -4000 -4425 -4900 -5425 -6000 -6625 -7300 -8025 40 -8800 -2500 -1800 -1525 -1000 -925 -900 -925 -1300 -1525 -1800 -2125 -2925 -3400 -4500 -5125 -5800 -6525 50 -2125 -1300 -1125 -1000 -1125 -2500 -3925 -7300 60 -1200 -825 -225 175 300 375 400 375 175 -225 -500 -825 -1200 -1625 -2100 -2625 -3200 -3825 -4500 -5225 -6000 -500 300 -250 350 750 950 975 750 350 -625 -2625 -4650 65 -625 75 575 875 950 875 575 75 -250 -1050 -1525 -2050 -3250 -3925 -5425 70 -100 600 875 1100 1275 1400 1500 1475 1275 1100 875 600 275 -1000 -2725 275 1475 1400 -100 -525 -1525 -2100 -3400 -4125 -4900 75 375 750 1075 1350 1575 1750 1875 1950 1975 1950 1875 1750 1575 1350 1075 750 375 -50 -52 -1050 -1625 -2250 -2925 -3650 -4425 375 80 800 1175 1500 1775 2000 2175 2300 2375 2400 2375 2300 2175 2000 1775 1500 1175 800 -100 -625 -1200 -1825 -2500 -3225 -4000 85 1175 1550 1875 2150 2375 2550 2675 2750 2775 2750 2675 2550 2375 2150 1875 1550 1175 750 27 -250 -825 -1450 -2125 -2850 -3625 90 1500 1875 2200 2475 2700 2875 3000 3075 3100 3075 3000 2875 2700 2475 2200 1875 1500 1075 600 75 -500 -1125 -1800 -2525 -3300 1775 2150 2475 2750 2975 3150 3275 3350 3375 3350 3275 3150 2975 2750 2475 2150 1775 1350 875 350 -225 -850 -1525 -2250 -3025 95 100 2000 2375 2700 2975 3200 3375 3500 3575 3600 357 3500 3375 3200 2975 2700 2375 2000 1575 1100 575 -625 -1300 -2025 -2800 2175 3675 3775 3750 2875 2550 1750 175 105 2550 2875 3150 3375 3550 3750 3675 3550 3375 3150 2175 1275 750 -450 -1125 -1850 -2625 3675 3875 3900 3675 3275 1875 875 2300 2675 3000 3275 3500 3800 3875 3800 3500 3000 2675 2300 1400 300 -325 -1000 -1725 110 -2500 3075 375 2375 2750 3075 3350 3575 3750 3875 3950 3975 3950 3875 3750 3575 3350 2750 2375 1950 1475 -250 -925 -1650 -2425 115 950 120 2400 2775 3100 3375 3600 3775 3900 3975 4000 3975 3900 3775 3600 3375 3100 2775 2400 1975 1500 975 400 -225 -900 -1625 -2400 2750 125 2375 3075 3350 3575 3750 3875 3950 3975 395( 3875 3750 3575 3350 3075 2750 2375 1950 147 950 375 -250 -925 -1650 -2425 130 2300 3800 3000 2675 3000 327 3500 3675 3875 3900 387 3800 3675 3500 3275 2675 2300 1875 1400 875 300 -325 -1000 -1725 -2500 135 2175 2550 2875 3150 3375 3550 3675 3750 3775 3750 3675 3550 3375 3150 2875 2550 2175 1750 1275 750 175 -450 -1125 -1850 -2625 2000 2375 3375 3500 3575 3600 3375 2975 2700 2375 1575 575 -625 -1300 -2025 140 2700 2975 3200 3575 3500 3200 2000 1100 -2800 3375 2475 145 1775 2150 2475 2750 2975 3150 3275 3350 3350 3275 3150 2975 2750 2150 1775 1350 875 350 -225 -850 -1525 -2250 -3025 150 1500 1875 2200 2475 2700 2875 3000 3075 3100 3075 3000 2875 2700 2475 2200 1875 1500 1075 600 75 -500 -1125 -1800 -2525 -3300 155 1175 1550 1875 2150 2375 2550 2675 2750 2775 2750 2550 2375 2150 1875 1550 1175 750 -825 -1450 -2125 -2850 2675 275 -250 -3625 160 800 1175 1500 1775 2000 2175 2300 2375 2400 2375 2300 2175 2000 1775 1500 1175 800 375 -100 -625 -1200 -1825 -2500 -3225 -4000 170 -100 275 600 875 1100 1275 1400 1475 1500 1475 1400 1275 1100 875 600 275 -100 -525 -1000 -1525 -2100 -2725 -3400 -4125 -4900 -1200 -825 -500 175 375 400 175 -225 -500 -1200 -1625 -3200 -3825 -4500 -5225 180 -225 300 375 -825 -2100 -2625 -6000 300 -1800 -1125 -925 -900 -1525 -1800 -2125 -2500 -2925 -4500 -5125 -6525 190 -2500 -2125 -1525 -1300 -1000 -925 -1000 -1125 -1300 -3400 -3925 -5800 -7300 -4000 -3625 -3300 -3025 -2800 -2625 -2500 -2425 -2400 -2425 -2500 -2625 -2800 -3025 -3300 -3625 -4000 -4425 -4900 -5425 -6000 -6625 -7300 -8025 -8800 200 -4100 -4325 -4725 210 -5700 -5325 -5000 -472 -4500 -4325 -4200 -4125 -4125 -4200 -4500 -5000 -5325 -5700 -6125 -6600 -7125 -7700 -8325 -9000 -9725 -10500 220 -7600 -7225 -6900 -6625 -6400 -6225 -6100 -6025 -6000 -6025 -6100 -6225 -6400 -6625 -6900 -7225 -7600 -8025 -8500 -9025 -9600 -10225 -10900 -11625 -12400 230 -9700 -9325 -9000 -8725 -8500 -8325 -8200 -8125 -8100 -8125 -8200 -8325 -8500 -8725 -9000 -9325 -9700 -10125 -10600 -11125 -11700 -12325 -13000 -13725 -14500 -12000 -11625 -11300 -11025 -10800 -10625 -10500 -10425 -10400 -10425 -10500 -10625 -10800 -11025 -11300 -11625 -12000 -12425 -12900 -13425 -14000 -14625 -15300 -16025 -16800 240 250 -14500 -14125 -13800 -13525 -13300 -13125 -13000 -12925 -12900 -12925 -13000 -13125 -13300 -13525 -13800 -14125 -14500 -14925 -15400 -15925 -16500 -17125 -17800 -18525 -19300 -15625 -16225 -17200 260 -17200 -16825 -16500 -16225 -16000 -15825 -15700 -15600 -15625 -1570 -15825 -16000 -16500 -16825 -17625 -18100 -18625 -19200 -19825 -20500 -21225 -22000 270 -20100 -18725 -18600 -18525 -18500 -18525 -18725 -19125 -19400 -19725 -21000 -21525 -22725 -1972 -19400 -1912 -18900 -18600 -18900 -20100 -2052 -22100 -23400 -24125 -24900 -23200 -22500 -22500 -22825 280 -22825 -22225 -22000 -21825 -21700 -21625 -21600 -21625 -21700 -21825 -22000 -22225 -23200 -23625 -24100 -24625 -25200 -25825 -26500 -27225 -28000

### Table A: Player A's Payoff in Points=4000-(Final Container 1 amount-120)<sup>2</sup>-(Final Container 2 amount -40)<sup>2</sup>

# Table B: Player B's Payoff in Points=4000-(Final Container 2 amount -f)<sup>2</sup>

orecast, f of		Amount in	Container 2	<u>→</u>																						
ontainer 2 Amt.		0	5	10	15	20	25	30	35	40	45	50	55	60	65	70	75	80	85	90	95	100	105	110	115	120
$\downarrow$	0	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625	-2400	-3225	-4100	-5025	-6000	-7025	-8100	-9225	-10400
	5	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625	-2400	-3225	-4100	-5025	-6000	-7025	-8100	-9225
	10	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625	-2400	-3225	-4100	-5025	-6000	-7025	-8100
	15	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625	-2400	-3225	-4100	-5025	-6000	-7025
	20	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625	-2400	-3225	-4100	-5025	-6000
	25	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625	-2400	-3225	-4100	-5025
	30	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625	-2400	-3225	-4100
	35	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625	-2400	-3225
	40	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625	-2400
	45	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900	-1625
	50	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225	-900
	55	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400	-225
	60	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975	400
	65	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500	975
	70	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975	1500
	75	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400	1975
	80	-2400	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775	2400
	85	-3225	-2400	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100	2775
	90	-4100	-3225	-2400	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375	3100
	95	-5025	-4100	-3225	-2400	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600	3375
	100	-6000	-5025	-4100	-3225	-2400	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775	3600
	105	-7025	-6000	-5025	-4100	-3225	-2400	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900	3775
	110	-8100	-7025	-6000	-5025	-4100	-3225	-2400	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975	3900
	115	-9225	-8100	-7025	-6000	-5025	-4100	-3225	-2400	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000	3975
	120	-10400	-9225	-8100	-7025	-6000	-5025	-4100	-3225	-2400	-1625	-900	-225	400	975	1500	1975	2400	2775	3100	3375	3600	3775	3900	3975	4000

# SFB 649 Discussion Paper Series 2016

For a complete list of Discussion Papers published by the SFB 649, please visit http://sfb649.wiwi.hu-berlin.de.

- 001 "Downside risk and stock returns: An empirical analysis of the long-run and short-run dynamics from the G-7 Countries" by Cathy Yi-Hsuan Chen, Thomas C. Chiang and Wolfgang Karl Härdle, January 2016.
- 002 "Uncertainty and Employment Dynamics in the Euro Area and the US" by Aleksei Netsunajev and Katharina Glass, January 2016.
- 003 "College Admissions with Entrance Exams: Centralized versus Decentralized" by Isa E. Hafalir, Rustamdjan Hakimov, Dorothea Kübler and Morimitsu Kurino, January 2016.
- 004 "Leveraged ETF options implied volatility paradox: a statistical study" by Wolfgang Karl Härdle, Sergey Nasekin and Zhiwu Hong, February 2016.
- 005 "The German Labor Market Miracle, 2003 -2015: An Assessment" by Michael C. Burda, February 2016.
- 006 "What Derives the Bond Portfolio Value-at-Risk: Information Roles of Macroeconomic and Financial Stress Factors" by Anthony H. Tu and Cathy Yi-Hsuan Chen, February 2016.
- 007 "Budget-neutral fiscal rules targeting inflation differentials" by Maren Brede, February 2016.
- 008 "Measuring the benefit from reducing income inequality in terms of GDP" by Simon Voigts, February 2016.
- 009 "Solving DSGE Portfolio Choice Models with Asymmetric Countries" by Grzegorz R. Dlugoszek, February 2016.
- 010 "No Role for the Hartz Reforms? Demand and Supply Factors in the German Labor Market, 1993-2014" by Michael C. Burda and Stefanie Seele, February 2016.
- 011 "Cognitive Load Increases Risk Aversion" by Holger Gerhardt, Guido P. Biele, Hauke R. Heekeren, and Harald Uhlig, March 2016.
- 012 "Neighborhood Effects in Wind Farm Performance: An Econometric Approach" by Matthias Ritter, Simone Pieralli and Martin Odening, March 2016.
- 013 "The importance of time-varying parameters in new Keynesian models with zero lower bound" by Julien Albertini and Hong Lan, March 2016.
- 014 "Aggregate Employment, Job Polarization and Inequalities: A Transatlantic Perspective" by Julien Albertini and Jean Olivier Hairault, March 2016.
- 015 "The Anchoring of Inflation Expectations in the Short and in the Long Run" by Dieter Nautz, Aleksei Netsunajev and Till Strohsal, March 2016.
- 016 "Irrational Exuberance and Herding in Financial Markets" by Christopher Boortz, March 2016.
- 017 "Calculating Joint Confidence Bands for Impulse Response Functions using Highest Density Regions" by Helmut Lütkepohl, Anna Staszewska-Bystrova and Peter Winker, March 2016.
- 018 "Factorisable Sparse Tail Event Curves with Expectiles" by Wolfgang K. Härdle, Chen Huang and Shih-Kang Chao, March 2016.
- 019 "International dynamics of inflation expectations" by Aleksei Netšunajev and Lars Winkelmann, May 2016.
- 020 "Academic Ranking Scales in Economics: Prediction and Imdputation" by Alona Zharova, Andrija Mihoci and Wolfgang Karl Härdle, May 2016.

## SFB 649, Spandauer Straße 1, D-10178 Berlin http://sfb649.wiwi.hu-berlin.de



This research was supported by the Deutsche Forschungsgemeinschaft through the SFB 649 "Economic Risk".

# SFB 649 Discussion Paper Series 2016

For a complete list of Discussion Papers published by the SFB 649, please visit http://sfb649.wiwi.hu-berlin.de.

- 021 "CRIX or evaluating blockchain based currencies" by Simon Trimborn and Wolfgang Karl Härdle, May 2016.
- 022 "Towards a national indicator for urban green space provision and environmental inequalities in Germany: Method and findings" by Henry Wüstemann, Dennis Kalisch, June 2016.
- 023 "A Mortality Model for Multi-populations: A Semi-Parametric Approach" by Lei Fang, Wolfgang K. Härdle and Juhyun Park, June 2016.
- 024 "Simultaneous Inference for the Partially Linear Model with a Multivariate Unknown Function when the Covariates are Measured with Errors" by Kun Ho Kim, Shih-Kang Chao and Wolfgang K. Härdle, August 2016.
- 025 "Forecasting Limit Order Book Liquidity Supply-Demand Curves with Functional AutoRegressive Dynamics" by Ying Chen, Wee Song Chua and Wolfgang K. Härdle, August 2016.
- 026 "VAT multipliers and pass-through dynamics" by Simon Voigts, August 2016.
- 027 "Can a Bonus Overcome Moral Hazard? An Experiment on Voluntary Payments, Competition, and Reputation in Markets for Expert Services" by Vera Angelova and Tobias Regner, August 2016.
- 028 "Relative Performance of Liability Rules: Experimental Evidence" by Vera Angelova, Giuseppe Attanasi, Yolande Hiriart, August 2016.
- 029 "What renders financial advisors less treacherous? On commissions and reciprocity" by Vera Angelova, August 2016.
- 030 "Do voluntary payments to advisors improve the quality of financial advice? An experimental sender-receiver game" by Vera Angelova and Tobias Regner, August 2016.
- 031 "A first econometric analysis of the CRIX family" by Shi Chen, Cathy Yi-Hsuan Chen, Wolfgang Karl Härdle, TM Lee and Bobby Ong, August 2016.
- 032 "Specification Testing in Nonparametric Instrumental Quantile Regression" by Christoph Breunig, August 2016.
- 033 "Functional Principal Component Analysis for Derivatives of Multivariate Curves" by Maria Grith, Wolfgang K. Härdle, Alois Kneip and Heiko Wagner, August 2016.
- 034 "Blooming Landscapes in the West? German reunification and the price of land." by Raphael Schoettler and Nikolaus Wolf, September 2016.
- 035 "Time-Adaptive Probabilistic Forecasts of Electricity Spot Prices with Application to Risk Management." by Brenda López Cabrera , Franziska Schulz, September 2016.
- 036 "Protecting Unsophisticated Applicants in School Choice through Information Disclosure" by Christian Basteck and Marco Mantovani, September 2016.
- 037 "Cognitive Ability and Games of School Choice" by Christian Basteck and Marco Mantovani, Oktober 2016.
- 038 "The Cross-Section of Crypto-Currencies as Financial Assets: An Overview" by Hermann Elendner, Simon Trimborn, Bobby Ong and Teik Ming Lee, Oktober 2016.
- 039 "Disinflation and the Phillips Curve: Israel 1986-2015" by Rafi Melnick and Till Strohsal, Oktober 2016.

## SFB 649, Spandauer Straße 1, D-10178 Berlin http://sfb649.wiwi.hu-berlin.de



This research was supported by the Deutsche Forschungsgemeinschaft through the SFB 649 "Economic Risk".

# SFB 649 Discussion Paper Series 2016

For a complete list of Discussion Papers published by the SFB 649, please visit http://sfb649.wiwi.hu-berlin.de.

- 040 "Principal Component Analysis in an Asymmetric Norm" by Ngoc M. Tran, Petra Burdejová, Maria Osipenko and Wolfgang K. Härdle, October 2016.
- 041 "Forward Guidance under Disagreement Evidence from the Fed's Dot Projections" by Gunda-Alexandra Detmers, October 2016.
- 042 "The Impact of a Negative Labor Demand Shock on Fertility Evidence from the Fall of the Berlin Wall" by Hannah Liepmann, October 2016.
- 043 "Implications of Shadow Bank Regulation for Monetary Policy at the Zero Lower Bound" by Falk Mazelis, October 2016.
- 044 "Dynamic Contracting with Long-Term Consequences: Optimal CEO Compensation and Turnover" by Suvi Vasama, October 2016.
- 045 "Information Acquisition and Liquidity Dry-Ups" by Philipp Koenig and David Pothier, October 2016.
- 046 "Credit Rating Score Analysis" by Wolfgang Karl Härdle, Phoon Kok Fai and David Lee Kuo Chuen, November 2016.
- 047 "Time Varying Quantile Lasso" by Lenka Zbonakova, Wolfgang Karl Härdle, Phoon Kok Fai and Weining Wang, November 2016.
- 048 "Unraveling of Cooperation in Dynamic Collaboration" by Suvi Vasama, November 2016.
- 049 "Q3-D3-LSA" by Lukas Borke and Wolfgang K. Härdle, November 2016.
- 050 "Network Quantile Autoregression" by Xuening Zhu, Weining Wang, Hangsheng Wang and Wolfgang Karl Härdle, November 2016.
- 051 "Dynamic Topic Modelling for Cryptocurrency Community Forums" by Marco Linton, Ernie Gin Swee Teo, Elisabeth Bommes, Cathy Yi-Hsuan Chen and Wolfgang Karl Härdle, November 2016.
- 052 "Beta-boosted ensemble for big credit scoring data" by Maciej Zieba and Wolfgang Karl Härdle, November 2016.
- 053 "Central Bank Reputation, Cheap Talk and Transparency as Substitutes for Commitment: Experimental Evidence" by John Duffy and Frank Heinemann, December 2016.



SFB 649, Spandauer Straße 1, D-10178 Berlin http://sfb649.wiwi.hu-berlin.de